

#### Technische Universität München Professur für Modellierung räumlicher Mobilität

#### MASTER'S THESIS

Evaluation of the Impact of Erding Ring Closure on Passenger Flows in Munich Region Evaluierung der Auswirkungen des Erdinger Ringschlusses auf Passagierströme in der Region München

> Author: Yonsorena Nong

Supervision: Hema Sharanya Rayaprolu Prof. Dr.-Ing. Rolf Möckel

Date of Submission: 2018-03-07

Technische Universität München – Professur für Modellierung räumlicher Möbilität



Prof. Dr.-Ing. Rolf Moeckel Arcisstraße 21, 80333 München, www.msm.bgu.tum.de

### **MASTER'S THESIS**

of Yonsorena Nong (Matriculation No.: 03676276)

Date of Issue: 2017-09-07

Date of Submission: 2018-03-07

#### <u>Topic:</u> Evaluation of the Impact of "Erding Ring Closure" on Passenger Flows in Munich Region

The Erding Ring Closure (Erdinger Ringschluss) is a proposed plan for passenger rail transport in Munich region. The regional planning authority of Munich region views this project as a key opportunity to improve the airport access, minimize the travel time, and provide direct commuter rail transport from Freising to Erding via Munich Airport. Upon completion, the newly constructed line of urban rail transit will have a series of influences on the whole city public transportation network. In this context, a study on travel time and passenger flow to Munich airport will be conducted. Since the new transit line will provide a direct connection between Munich airport and Messe, the travel time between these stations will also be assessed. Furthermore, an assessment of passenger flow on Stammstrecke will be made as it will have new characteristic due to the new transit line going into operation.

To investigate the effect of Erding Ring Closure, a passenger flow prediction framework will be built and the prediction model will be constructed. The simulation will be conducted and performed using VISUM. Moreover, the results from the simulations will be analyzed to predict the changes in mode shift and passenger volume of a new line of urban rail transit.

The research conducted in this thesis will help deliver a better insight into the future travel demand in Munich region. Finally, the results from the comparison between the base scenario and the scenarios with the Erding Ring Closure will help us identify its effect.

The student will present intermediate results to the supervisors (Prof. Dr.-Ing. Rolf Möckel and Hema Sharanya Rayaprolu) in the fifth, tenth, 15th and 20th week.

The student must hold a 20-minute presentation with a subsequent discussion at the most two months after the submission of the thesis. The presentation will be considered in the final grade in cases where the thesis itself cannot be clearly evaluated.

Prof. Dr.-Ing. Rolf Moeckel

Nim

Student's signature

Hema Sharanya Rayaprolu

#### Abstract

With an increasing awareness of sustainability, most major European cities have invested in public transport to tackle increasing congestion, reduce high reliance on automobile, and reduce daily mobility costs. Studies revealed that making changes to infrastructure or overall quality of service of public transport may have significant impact on daily mobility of individuals and transit ridership. One anticipated project discussed in this thesis is Erding Ring Closure, a new rail connection between Erding and Munich Airport. As the project involves large scale development, feasible study of its impact is deserved.

This thesis aims to investigate the potential effects of the Erding Ring Closure on passenger flows in Munich region. To realize the impact of the project on individual mobility behavior as a consequence of infrastructure and quality service improvement, a discrete mode choice model and a route choice model are built. The study explicitly includes both short and long-distance travel model within a single model framework. While the trip to airport is modeled based on long-distance travel model, the urban travel model is implemented for commuting and other trip. Only two alternatives are considered in the study - auto and transit.

The study determines the influence of new opening service on mode choice using a binary logit model. Constrained by the lack of data available, derived parameters from previous research are adopted to formulate mode choice model. Equilibrium assignment and timetable-based approach are applied to allocate the mode-specific trips on the routes and links. PTV VISUM, a macro traffic flow simulation, is used to facilitate the analysis process of mode choice and route assignment. Three scenarios are modeled to predict the changes in modal split as well as to test the model sensitivity. Those scenarios are scenario with Erding Ring Closure and the other two scenarios with increasing auto operating cost and parking cost after Erding Ring Closure is put into service.

Model results of scenario with Erding Ring Closure provides evidence that additional transit service has not been sufficient to have a substantial impact on modal split in Munich region, although there has been less of a reduction in auto and slower growth in transit in comparison to base scenario. The minimal positive effects of Erding Ring Closure may indicate travel time is still the most important factor that affect individual decision on travel mode. Another two scenarios tell a very different story. The scenario with high operating cost and parking cost shows a growth in transit share regardless of trip purposes. Besides, the study fails to perform and analyze the assignment task as unrealistic results is yield from the model. The study concludes by addressing the impact of Erding Ring Closure on travel mode share.

#### Acknowledgment

First of all, I would like to express my profound gratitude to my advisor, Hema Sharanya Rayaprolu for her continual advice, support, and her trust in my ability. Her guidance and advice are invaluable and indispensable. Thank you for your patience, encouragement, and immense knowledge you gave me within this 6 months.

Secondly, I would like to thank my advisor, Prof. Dr.-Ing Rolf Moeckel for approving and supporting the idea of this thesis. Thank you for your great advice during the development of the thesis.

Beside my advisors, I would like to thank Ms. Demmel Tatjana from Flughafen München GmbH for her support of this thesis by providing me with data needed to facilitate the research.

Furthermore, I would like to thank PTV AG for supporting my research by providing me with the software and always responding to my questions about modeling in VISUM.

I would like to further extend my thank to Dr. Carlos Llorca and Dr.-Ing. Matthew Bediako Okrah. I am highly indebted to their help and advice along the way.

Finally, I would like to express my sincere gratitude toward my family, especially my parents who unconditionally support me and always love me for what I want to pursue. Also, this thesis would have never been possible without the constant support from my sisters and my friends from all around the world.

#### Declaration concerning the Master's Thesis

I hereby confirm that the presented thesis work has been done independently and using only the sources and resources as are listed. This thesis has not previously been submitted elsewhere for purposes of assessment.

Munich, March 7th, 2018

Yonsorena Nong

# Contents

A	bstra	i			
$\mathbf{A}$	Acknowledgment ii				
D	Declaration concerning the Master's Thesis iii				
$\mathbf{C}$	onter	iv			
$\mathbf{Li}$	st of	Figures vi			
$\mathbf{Li}$	st of	Tables viii			
$\mathbf{Li}$	st of	Abbreviations ix			
1	<b>Intr</b> 1.1 1.2 1.3	oduction       1         Background of Erding Ring Closure       2         Objective       3         Outline       3			
2	Lite 2.1 2.2 2.3	rature Reviews4Impact of new transit infrastructure42.1.1 Remark5Four-Steps Model6Mode Choice72.3.1 Factors influencing the choice of mode72.3.2 Discrete Choice Theory72.3.3 Linear Probability Model82.3.4 Probit Model92.3.5 Logit Model92.3.6 Other Models10Airport choice model10			
	2.5 2.6	2.4.1Heuristically derived parameters112.4.2Remark12Route Assignment122.5.1All-or-Nothing122.5.2Capacity-restraint Assignment122.5.3Incremental Assignment132.5.4Wardrop's First Principle - User Equilibriums132.5.5Wardrop's First Principle - Stochastic User Equilibrium14			
		2.6.1       Overview of VISUM 17       14         2.6.2       Mode Choice in VISUM       14			

		2.6.3	Route Assignment in VISUM	15	
3	Met	Methodology 1			
4	<b>Dat</b> 4.1 4.2 4.3	a Anal Genera 4.1.1 4.1.2 Genera Geospa	ysis and Processing         ating OD Matrices         Incorporating exogenous data         Assessing OD Matrices         al Transit Feed Specification (GTFS)         atial data	<ol> <li>19</li> <li>20</li> <li>20</li> <li>21</li> <li>21</li> </ol>	
5	<b>Stu</b> 5.1 5.2 5.3 5.4	dy Are Selecti Curren 5.2.1 5.2.2 Analys Transp 5.4.1	a         on of study area         nt situation of urban mobility         Road Network         Public Transport Network         sis of travel times         ort Analysis Zone System         Airport Zone System	<ul> <li>23</li> <li>23</li> <li>25</li> <li>28</li> <li>29</li> <li>29</li> <li>31</li> <li>33</li> </ul>	
6	Moo 6.1 6.2 6.3	deling Model 6.1.1 6.1.2 6.1.3 Model 6.3.1 6.3.2 6.3.3 6.3.4 6.3.5 6.3.6 6.3.7 6.3.8	Framework         Specification	$\begin{array}{c} {\bf 34}\\ {\bf 34}\\ {\bf 34}\\ {\bf 36}\\ {\bf 37}\\ {\bf 38}\\ {\bf 41}\\ {\bf 41}\\ {\bf 41}\\ {\bf 42}\\ {\bf 43}\\ {\bf 44}\\ {\bf 44}\\ {\bf 45}\\ {\bf 48} \end{array}$	
7	Scen 7.1 7.2 7.3 7.4 7.5 7.6	nario A Public Predic Scenar Scenar Scenar Summa	Analysis         transport network	<ul> <li>49</li> <li>49</li> <li>50</li> <li>50</li> <li>52</li> <li>53</li> <li>54</li> </ul>	
8	Con 8.1 8.2 8.3	Conclu Conclu Limita Sugges	n uding remark	<b>56</b> 57 57 58	
$\operatorname{Bi}$	bliog	graphy		<b>59</b>	

# List of Figures

1.1	Erding Ring Closure
$2.1 \\ 2.2$	Standard Four-steps models    6      Graph of CR-Function with different parameter of b    13
$3.1 \\ 3.2$	Proposed Methodology Framework    17      Proposed Four-steps model for the study    18
4.1	OSM Data Manipulation Task
$5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5 \\ 5.6 \\ 5.7 \\ 5.8 \\ 5.9 \\ 5.10 \\ 5.11 \\ 5.12 $	Study area for this research24Modal Split in districts within study area26Number of car per household26Car availability as a driver27Accessibility to workplace with transit27Private network within study area28Public Transport in districts within study area29Transit travel time to airport in minutes30Car travel time to airport in minutes31Traffic zones of study area33Zone system of airport33
5.12 6.1	Zone system of airport
6.2	Network modeling in VISUM
6.3	An example of network in city center
6.4	An example of network in airport area
6.5	Simplified flowchart of mode choice model in VISUM
6.6	Comparison between actual and predicted share of auto
6.7	Comparison between actual and predicted share of transit
6.8	Simplified flowchart of route assignment model in VISUM
6.9	An example of PrT assignment result in VISUM - Threshold of 5,000 Pers trip with scale dimension 2
6.10	An example of PuT assignment result in VISUM - Threshold of 5,000 Pers trip with scale dimension 2
6.11	Assignment results in base scenario - Threshold of 500 Pers Trip with scale dimension 1.5
7.1	Erding Ring Closure Scenario
7.2	Air passenger mode share - Comparison between base scenario and Erding Ring Closure Scenario

7.3	Mode share to airport - Comparison between base scenario and Erding Ring	
	Closure Scenario	51
7.4	Mode share in study area - Comparison between base scenario and Erding Ring	
	Closure Scenario	52
7.5	Air passenger mode share - Comparison between base scenario and Doubled AOC	52
7.6	Mode share to airport - Comparison between base scenario and Doubled AOC	53
7.7	Mode share in study area - Comparison between base scenario and Doubled AOC	53
7.8	Mode share to airport - Comparison between base scenario and Doubled PC	54
7.9	Mode share in study area - Comparison between base scenario and Doubled PC .	54

# List of Tables

4.1	Summary of trips by purpose	9
6.1	Variable considered in the estimation	6
6.2	Coefficients for utilities of auto	7
6.3	Coefficients for utilities of transit	7
6.4	Constant for auto and transit	8
6.5	Summary of inputs required to run model	9
6.6	Result of a test for accuracy of model fit for Homebased Other	3
6.7	Result of a test for accuracy of model fit for Homebased Work 4	3
6.8	Result of a test for accuracy of model fit for Homebased Airport 4	4
6.9	Average vehicle occupancies according to trip purpose	4
7.1	Summary of new transit service for all scenario	9
7.2	Mode share for trips in Munich region by scenario	5

# List of Abbreviations

ADAC	Allgemeiner Deutscher Automobilclub e.V
AOC	Auto Operating Cost
HA	Home-Airport
НО	Home-Other
HW	Home-Work
MiD	Mobilität in Deutschland
MVG	Münchner Verkehrgesellschaft
MVV	Münchner Verkehr und Tariffverbund
OD	Origin Destination
PrT	Private Transport
PuT	Public Transport
PC	Parking Cost
РКТ	Passenger Kilometer Travel
TAZ	Transport Analysis Zone
VDV	Vereband Deutshcer Verkehersunternehmen
VMT	Vehicle Mile Travel

## Chapter 1

## Introduction

Special attention has been given to public transport because its role in transportation has a huge impact on the quality of life in the society. As considered by many, public transport closes a gap between social and economic mobility, and improves ecological sustainability (Eichmann, Berschin, Bracher, & Winter, 2005). There is a mounting evidence that the rapid development of urban rail transit is taken place due to the current traffic congestion, increasing environmental issues, and the heavy load of private transport on the road.

In Germany, there are many types of mass rapid transit in operation, ranging from bus to subway. Noticeably, one of rapid transit types is suburban rail systems called S-Bahn or Schnellstadtbahn, the hybrid urban-suburban type which operates in almost all major metro areas. S-Bahn serves and connects the urban as well as suburbs to many regional towns. Like other rail transits around the world, S-Bahn system in Germany is well-established and a very well-integrated public transport network. Some of the good model networks can be found in Munich, Berlin, or Rhein-Ruhr.

In the last two decades, intensive upgrade and expansion of the mass transit system in Germany have been undertaken to acknowledge the needs for a better and convenient mobility. One of the major rapid transit project is Erdinger Ring Closure (German: *Erdinger Ringschluss*), a proposed new rail connection between Erding and Munich Airport. This new development has attracted not only the attention of public, but also traffic planners and researchers. The government and project developer believe that the project is a promising development that aims to promote local transport within the regions. Moreover, this development also emphasizes a clear tendency of the regional planning authority in Munich region regarding the improvement of public transport. The development of this project could make the urban rail transit and regional rail transit more attractive. Also, it improves the accessibility of certain destinations in Erding and other regions in the northeast of Munich.

There is evidence that improvement of the infrastructure or the services usually have a large impact on transit ridership. Munich's public transport network is expected to have a new characteristic on both passenger flows and distribution on the entire network when the new S-Bahn service is put into operation. In this case, the new line route will impose a significant impact on the network, such as causing changes in passenger volumes and distribution features on other line routes. It is therefore necessary to investigate the growth patterns of traffic network and assess the influence of new service.

#### 1.1 Background of Erding Ring Closure



Figure 1.1: Erding Ring Closure Source: Openstreetmap and elaborated by Author with ArcGIS

The Erding Ring Closure (German: Erdinger Ringschluss) is a proposed passenger rail transport project in Erding (see Figure 1.1). It is the second major rail transport development after 2nd Trunk Tunnel (German: 2nd Stammstrecke) which has been approved by government of Free State of Bavaria. Upon completion, an exclusive S-Bahn service will be operated between Freising and Erding via Munich Airport. In addition to S-Bahn service, two new additional service will also be put into service to serve airport bound trips e.g. trans-regional airport express (German: Überregionaler Flughafenexpress, ÜFEX) and airport express (German: Flug hafenexpress, FEX) (Bayerishes Staatministerium des Innern, für Bau und Verkehr [STMI], 2010).

For decades, there is an ongoing discussion on a connection between Erding and Munich Airport, which would be an important alternative to the existing transit connections to the airport. At the same time, this connection is also a prerequisite for a better connection to southeastern Bavaria. At present time, there is a missing rail connection between Erding and airport, and therefore a direct connection between Erding and Freising as well. Commuters from Erding to airport or Freising have to use modes other than rail transit (e.g., bus, car-sharing, or private vehicles). Journey with bus would be less convenient and comfort especially for passengers carrying big and a lot of luggage. On the other hand, traveling with personal private vehicles would be much better in most situations, but however with higher mobility cost. Therefore, this development is a key opportunity to improve the airport access, minimize the travel time, and provide direct commuter rail transport from Freising to Erding via Munich Airport. Apart from the airport bound access, the ring closure also plays an important role in relieving the effects on other lines and traffic load in Munich's city center. Under this circumstance, visitors for the fair in Messestadt would no longer have to travel via Hauptbahnhof or Ostbahnhof as the Messestadt of Riem would be directly connected to the airport.

#### 1.2 Objective

The purpose of the thesis is to address the characteristics of passenger flows before and after the new transit service on Erding Ring Closure is introduced. The study is carried out based on the conventional four-steps model by using VISUM, an integrated transport planning software. In this research, a mode choice model is built to help formulating and estimating modal split and subsequently analyzing route assignment in the later stage. The goal of the thesis can be achieved through this research question:

How does the Erding Ring Closure affect the mode choice and route choice of a number of trips in Munich Region?

In this thesis, I have divided the tasks into several steps, which are a critical stepping stone in leading this research toward the goal. The following tasks are carried out to support answering the research question:

- 1. Predicting the number of trips based on individual mode (Auto and Transit)
- 2. Building mode choice model for the study
- 3. Analyzing and formulating the trip assignment procedure
- 4. Forecasting the mode choice and route choice under the scenario "Erding Ring Closure", scenario doubled auto operating cost, and scenario doubled parking cost

The research conducted in this thesis will deliver a better insight into the future travel demand in Munich region, especially the northeasten part of Munich and Erding. Finally, the results from the comparison between the base scenario and the scenarios with the Erding Ring Closure will help us identify its effect. The results from this study can also be used to support other future researches related to airport demand modeling.

#### 1.3 Outline

The thesis is organized into 8 Chapters as follows:

- Chapter 2 discusses mainly the relevant literature in general for this thesis. The chapter begins with the review of previous studies and further describes the theory and approach of mode choice and route assignment.
- Chapter 3 outlines the methodology used in this thesis and briefly explains its procedures.
- Chapter 4 details the data used to build the model.
- Chapter 5 describes how the study area is selected and discusses the current situation provided with relevant information.
- Chapter 6 explains the modeling framework which includes the discussion of model specification, building, and estimation.
- Chapter 7 focuses on the comparison between based scenario and scenario with Erding Ring Closure according to the model built in chapter 6.
- Chapter 8 summarizes the study with a reflection, and outlines the limitations of the study with a suggestion for future researches.

## Chapter 2

## Literature Reviews

The chapter of Literature review is structured into two major parts. The first part focuses on the relevant literature (previous researches with similar point of interests). As several studies have already been conducted to analyze impacts of new transit lines in urban areas, the review of these studies will provide me some initial clues on the study results to be expected. The second part of this chapter briefly discusses about the theoretical literature (e.g., four-steps model, mode choice modeling, and route assignment). Also in this part, principal theory of four-steps model and its comprehensive application in the scope of transport system analysis will be discussed. Then it outlines the third and fourth step of the conventional four-steps model by providing the basic theory and its practical application with a concise explanation.

#### 2.1 Impact of new transit infrastructure

Undeniably, adding new transit lines, adding a highway lane, or building a complete new highway will change the network topology. Many studies have been conducted to investigate the correlation between urban mobility and performance of transport supply by analyzing the changes in infrastructure and services. Through empirical analysis, there is evidence that expanding transit infrastructure, improving transit services, and making cities friendly to non-motorized transports (i.e., pedestrian and bike) can directly affect individual mobility behavior.

Li (2015) noted that on grounds of a new transit line, the current passenger flows are like to change because of the new network structures. To a greater extend, Li explained that the expansion of the network will likely pose as an important significance on flow generation and distribution on each line. Likewise, Zheng (2008) also stated that the characteristic of passenger flow will change when the network connectivity is improved. Zheng conducted a study on the impact of new rail transit line on passenger flow in Shanghai and systematically summarized that passenger volume on each line and traveler's mobility choice are influenced by the enhancement of the connectivity of the whole network. For instance, passenger favors the path with shorter travel time and less transfer.

In a paper, Nixon et al., (2015) examined the changes in travel behavior of travelers in Los Angeles before and after the opening of new light rail transit service (LRT). They conducted a travel survey along Exposition and Crenshaw corridor where LRT will be operated along side the bus. Interestingly, they found that the newly introduced LRT has negative effects on overall transit ridership. For instance, bus ridership saw a dramatic reduction mainly due to the service changes. Furthermore, the study indicated that there is significant decrease in driving (vehicle miles travel [VMT]) of households who live within 1 km buffer zone of new LRT station. In contrast, those households living near bus stops removed increased their driving because of new LRT service. In another study, Li (2015) investigated if the additional of new urban rail transit line in Beijing will influence the passenger flow distribution in the network. The study found that the changes of passenger flow characteristics caused by the opening of different kinds of new subway lines can result in convenient transfer as well as increasing in passenger volume. According to the study, the current network alters significantly when new urban trunk line and supplementary loop line are in service. Opening of urban trunk line will increase the passenger volume of most transfer stations while new loop line (outer ring) will reduce passenger volume on the existing inner ring line. However, the study did not disclose the shift in mode choice between private and public transport. On top of that, Li emphasized that the land-use patterns and sub-urbanization process are directly affected by transport supply improvement. On these terms, it refines the land development intensity and widens passenger perception on rail transit.

Djoko (2014) analyzed the significant impact of proposed new metro line in Stockholm on the public transport network in his study. Even though his study focused only on the public transport (metro and bus), the study found interesting results with respects to travel time saving and passenger load. With the new metro line, travel times from Danderyds sjukhus station to Älvsjö station was notably reduced compared to current situation. Distrubtion of passenger loads on this north-south service also changed. The analysis saw a sharp decrease in occupancy rate on the bus line because the new metro would attract long distance passengers from the bus. Moreover, the new metro line was also able to relieve the congestion at the most crowded station in the network.

Besides, Siemen AG (2012) studied the impact of new public transport mode, LRT in Turku. The study was conducted to assess any beneficial impacts of LRT on society, ecology, and environment. The study compared the base scenario with the scenario with LRT that will be implemented in 2025 and 2035. Interestingly, the study found that the addition of LRT will decrease the car share in the modal split. The study further pointed out that LRT will partially replace the bus and attract car users. Moreover, passenger kilometers travel (PKT) with public transport also experiences positive increase with the implementation of LRT.

#### 2.1.1 Remark

Results from above papers may not be as homogeneous as it seems. This might be due to the pattern of transport network, location of study area, different travel behavior, and other associated factors. Some studies found that the shift from one alternative to another alternative has been wide as expected while some suggested that the new transit line did not add any competition to another modes. However, those studies have provided evidence that the change in network pattern can result in inducing more demand. In particular, better accessibility to a certain destination means more people will travel to perform the activity there.

#### 2.2 Four-Steps Model

Four-steps travel demand models are a traditional travel demand modeling procedure, primarily used to forecast the future demand and performance of a transport system. Four-steps model (FSM) is a simplified representation of reality which is suitable for large scale infrastructure project and not appropriate for subtle and complex policies that involve the analysis of individual travel behavior. FSM is a trip based model that perform the analysis based on the number of trips and estimate travel with different means of transport at the aggregate zonal level (McNally, 2007).

Normally, calculation for entire model of FSM involves four steps in sequence, and each step serves different purposes that would subsequently build up a model called four steps model. As defined by various traffic analysts and expertise, the four steps of the classical four-step models are:

- **Trip Generation**: determines the number of trips attracted and produced in each zone by trip purpose, as a function of land uses and household demographics, and other socioeconomic factors.
- **Trip Distribution**: matches the attraction and production trips between all zones of each trip purpose. The estimation is often measured with travel impedance (time and/or cost).
- Mode Choice: factors the trips and computes the proportion of trips for each particular mode of transport.
- **Route Assignment**: trips between origin and destination are distributed to mode specific network. Several factors such as minimum travel time and/or congestion are considered to determine the traffic volumes.



Figure 2.1: Standard Four-steps models Source: Adapted from EPA420-R-97-007, 1997

#### 2.3 Mode Choice

To clarify the question of what impact the infrastructure measures have on the mode choice, the definition of mode choice must first be defined. Modal split is the distribution of transport demand among the different modes of transport (Kirchhoff, 2002). It arises from the fact that people decide for their paths to be traveled with a particular mode of transport. In transport planning, mode choice is one of four sequential planning stage. The third step of four-steps model effectively factors the trip matrices from trip distribution to allocate the incurred trips on the existing modes. This analysis allows modeler to determine the proportion of existing modes on network and reflect the choice probabilities of individual trip makers.

#### 2.3.1 Factors influencing the choice of mode

The traveler's preference on alternative choice set is notably influenced by many various factors, ranging from transport-specific factors (describing the various parts of the transport system) to individual-related factors, namely car ownership or income (Olsson, 2003). Ortúzar and Willumsen (2011) classified these factors as follows:

- Characteristics of individual: relates to not only personal attributes of trip maker, but also to a certain degree of household, for example, car availability, possession of driving license, size of household.
- Characteristics of journey: relates to the trip itself, for example, trip purpose, time of the day.
- Characteristics of transport facilities: relates to the various components of transport system, for example, timetables, relative travel time, monetary travel cost, parking cost.

#### 2.3.2 Discrete Choice Theory

Earlier development of mode choice model was more concerned with the usual analytic approach, aggregate travel demand. Traffic planners often used aggregate approach to demonstrate the travel demand forecasting process. Travel data are often subdivided into zones, called travel analysis zones (TAZs) prior to the determination of travel patterns (including intensity of land-use and socioeconomic of population) (Weiner, 1997, p. 42). However, the focus on econometric and psychometric modeling has led to a major development of disaggregate travel demand models. Currently, the development of mode choice model is more concerned with the disaggregate level which is based on the discrete choice analysis methods. In contrast to the aggregate approach, Koppelman and Bhat (2006) stated that the disaggregate model is used more often because it can reflect an individual's choice on one alternative among a finite set of alternatives. To some extend, the model is built based on the individual attribute related to trip makers rather than statistical associations of a larger group (Ji, 2017).

The discrete choice model is an empirical model derived from utility theory. The model uses the principle of utility maximization to explore the choice of individual. The widely adopted function of utility maximization gives us an initial clue that an individual chooses the alternative that has the highest utility among a choice set (Schiffer, 2012, p. 32). This also means that individuals with the same attribute values and similar socio-economic characteristics would always select the same alternative. However, this is unrealistic because similar individuals can still choose different alternatives. To address these irrationalities, some random components are added to the utility function, and thus makes utility model random as well. Utility function can then be formulated through a combination of systematic components  $V_{in}$  and random components (disturbances)  $\varepsilon_{in}$ .

The equation is as follows:

$$U_{in} = V_{in} + \varepsilon_{in} \tag{2.1}$$

(Equation source: Ben-Akiva and Lerman, 1985, p. 60)

where:

 $U_{in}$ : utility of an alternative *i* to the decision maker *n* 

 $V_{in}$ : systematic components of utility *i* 

 $\varepsilon_{in}$ : random variable of error portion of the utility

The observable portion of the utility  $V_i$  is a function and assumed to be deterministic which is expressed by the following equation:

$$V_{in} = \beta_1 x_{in1} + \beta_2 x_{in2} + \beta_3 x_{in3} + \dots + \beta_K x_{inK}$$
(2.2)

(Equation source: Ben-Akiva and Lerman, 1985, p. 63)

where:

 $x_{in1}, x_{in2}, x_{in3}, \dots, x_{inK}$ : the independent variables that include both attributes of the alternative *i* and socioeconomic variables of the individual *n* 

 $\beta_1, \beta_2, \beta_3, \dots, \beta_K$ : the unknown parameters

Parameters  $\beta$  for each attribute can be asserted with Maximum Likelihood Estimation (MLE). The random component  $\varepsilon$  is basically represented by a probability distribution. However, it is important to note that different assumptions about the distribution of  $\varepsilon$  can result in different type of discrete choice models (Ben-Akiva & Lerman, 1985, p. 65).

Considering the choice set containing only two alternatives i and j, the following models are applicable for the binary case. The models described in the following sections depict discrete decisions, that is, behavioral choice with a finite number of distinguishable options. A related point to consider is that some of the decisions are due to the economic theory of utility maximization, as well as that some of the random decisions, or not to be fully replicated. The models presented differ in the way that they assume different distributions for the random error component. The simplest way is to assume that all random parts of a model follow the same distribution and are independent of each other. This property is called i.i.d (independent and identically distributed) (Lohse & Schneider, 1997).

#### 2.3.3 Linear Probability Model

The Linear Probability Model (LP model) is perhaps the simplest model compared to other models. The model is rooted in the assumption that the difference in disturbances,  $\varepsilon_{jn} - \varepsilon_{in}$  is uniformly distributed between two fixed values -L and L, with L>0. In LP model, the difference of  $\varepsilon_{jn} - \varepsilon_{in}$  and its density function are defined as  $\varepsilon_n$  and  $f(\varepsilon_n)$  respectively. The probability of choice in LP model can be solved according to the following equation:

$$P_n(i) = Pr(\varepsilon_n \le V_{in} - V_{jn}) \tag{2.3}$$

(Equation source: Ben-Akiva and Lerman, 1985, p. 67)

According to Ben-Akiva and Lerman (1985), the probability of choice for alternative *i* is given by the cumulative distribution function  $\varepsilon_n$ . When *V* is linear in its parameters, the probability function is also linear in the interval -L and L.

Despite being simple to use, this model has a major drawback. Ben-Akiva and Lerman (1985) described the results from the forecasts of LP model as unrealistic due to its "kinks" at the point L and -L. They further emphasized that the model fail to exclude some people who choose alternative i which is predicted by the model with a probability of 0. Another drawback of the model is the specific assumption on interval between -L and L, and zero probabilities outside this interval, and thus makes the model unrealistic (Cox, 1970). Although the calculation is relatively straightforward here, the model is rarely practiced due to its limited application for modeling with a larger set of alternatives, its problematic theoretical foundations, and the existence of better models in transport planning (Schnabel & Lohse, 2011).

#### 2.3.4 Probit Model

In probit model, it is assumed that the random components in utility function can be approached by the sum of a large number of unobserved, but independent components. An important aspect discussed here is the assumption about the distribution of  $\varepsilon_{in}$  and  $\varepsilon_{jn}$ . In the form of central limit theorem, the disturbances are normally distributed. Under the assumption of normal distribution for random component, the probability of choice for alternative *i* is given by:

$$P_n(i) = \Phi\left(\frac{V_{in} - V_{jn}}{\sigma}\right) \tag{2.4}$$

(Equation source: Ben-Akiva and Lerman, 1985, p. 69)

where:

- $\Phi$  : the standardized cumulative normal distribution
- $\sigma$ : the scale of utility function, usually  $\sigma=1$

In this model, the probability of choice only depends on  $\sigma$ , and even if the choice for  $\sigma$  is arbitrary, rescaling the  $\sigma$  and  $\beta$  by any positive constant would not affect the choice probability (Ben-Akiva & Lerman, 1985, p. 69).

The distribution function has a sigmoidal shape, in which the probability of choice is never zero or one. It approaches one when V is close to  $+\infty$  and zero when V is close to  $-\infty$ . The normal distribution with these properties is the most realistic representation of the stochastic utilities. However, this model is no longer practicable with more than two alternatives since the model cannot be calculated analytically. Thus, it is not very common in traffic planning (Schnabel & Lohse, 2011). In addition, the model is not considered convenient analytically due to its property of not having closed form (Ben-Akiva & Lerman, 1985, p. 70).

#### 2.3.5 Logit Model

Logit and probit models are very much alike, but what makes logit different from probit is the integral for the choice probability has a closed form. In terms of this property, logit model is considered analytically more convenient. Logit model relies on the assumption that the difference in disturbances  $\varepsilon_n$  is logistically distributed (Ben-Akiva & Lerman, 1985, p. 71). By virtue of this assumption, it means that all individual error component  $\varepsilon_{in}$  and  $\varepsilon_{jn}$  is distributed independently and type I extreme value (identically Gumbel distributed).

Assuming Gumbel Distribution for  $\varepsilon_n$ , the probability that individual choosing alternative *i* is expressed according to binary case as follows:

$$P_n(i) = \frac{e^{\mu \ V_{in}}}{e^{\mu \ V_{in}} + e^{\mu \ V_{jn}}}$$
(2.5)

(Equation source: Ben-Akiva and Lerman, 1985, p. 71)

where:

 $\mu$ : the positive scale parameter

For convenience, Ben-Akiva and Lerman (1985) suggested that  $\mu$  should be assumed to be one, similarly to the  $\sigma=1$  in probit model. However, it is worth noting that the standard scaling parameter for binary probit corresponds to the  $var(\varepsilon_{jn} - \varepsilon_{in})=1$ .

Even though logit model is widely used in comparison to other models, this model is incapable of capturing the correlation between alternatives. Ortúzar and Willumsen (2011) emphasized that logit model has the important IIA property (Independence from Irrelevant Alternatives property), which implies that the choice probabilities of two alternatives are not influenced by the existence of other alternatives. In reality, however, alternatives are not completely independent. A well-known example for it is red-bus and blue-bus.

#### 2.3.6 Other Models

Other common types of mode choice models is multinomial logit model (MNL), which is to some extend an extension of binary logit model. MNL is commonly used in the case when more than two alternatives are available in the network. Apart from MNL, the nested logit model (NL) is developed to deal with the major limitation of MNL Model. Basically, the NL model is an extension application of the MNL model whose decision is made in sequential process (Forinash & Koppelman, 1993, p. 99). As with the name itself, NL divides the choice alternatives into so-called nests. The model is rooted with the assumption that the random error terms are shared among the alternatives (Koppelman & Bhat, 2009, p. 159). The alternatives that are grouped in a nest are independent of each other. Hence, the IIA property is also within the nests. Between the nests, the probabilities of chosen alternative may depend on the existence or characteristics of other alternatives. The nested logit model assumes Gumbel distribution for the error terms,  $\varepsilon$  for each alternative whose the value lies between 0 and 1. Moreover, the model further assumes that the independent portion of the random variables for the nested alternatives are still independently distributed, but with a scale factor between 0 and  $\theta$  (Forinash & Koppelman, 1993; Koppelman & Bhat, 2009).

#### 2.4 Airport choice model

Due to the scope of study that explicitly includes the airport trip within a single model framework, the review of airport choice model is favored in this thesis. Airport choice model is often modeled based on long distance travel model. The reason for it is because majority of travelers to airport are long-distance traveler and they tend to stay longer at their destination than short-distance travelers. Airport trips tend to be done by auto or transit. On the other hand, travel mode like bike is not taken into account though it is possible that airport employee will bike to work. As cited in Moeckel et al. (2015), the following long-distance mode choice models offers the discussion of airport choice.

Blackstone et al. (2006) conducted a survey for airport choice from Baltimore to New York, that includes the major airports BWI, EWR, JFK, and PHI. Through their empirical analysis, the results presented in the study indicated that location of the workplace has a crucial impact on the decision process. For instance, business travelers with air decides on which airport to be used based on their workplace.

In another study, Başar and Bhat (2004) analyzed airport choice through their developed probabilistic choice set multinomial logit model. The study was set in the San Francisco Bay Area, where three major airports SFO, SJC, and OAK serve. The study found that frequency of service is also an important explanatory variable in addition to the access time.

Besides Blackstone et al. (2006) and Başar and Bhat (2004), Hess et al. (2007) analyzed the results from stated preference survey. The model built was estimated from the attribute that can be directly observed such as air fare, access time, or number of transfer. The analysis highlighted the significant effect of ground-level distance on airport-choice behaviour.

Other similar studies on airport choice can be referenced to the previous studies by Pel and Nijkamp (2001), Pel and Nijkamp (2003), Hess and Polak (2005), and Hess and Polak (2006a,b). Those studies adopted more sophisticated choice models such as nested logit model, cross-nested logit model, and mixed multinomial logit model.

In contrast to most other studies, Moeckel et al. (2015) developed a nested  $R^3$  logit model for long distance travel model in North Carolina. Unlike other studies that focus on airport choice, Moeckel et al. adopted station choice in their study for non-air modes (i.e., bus and rail). This approach seems to be more rational because the each major areas in North Carolina has only one major airport.

Despite the increasing importance of long-distance travel in transport modeling, only a few studies on long-distance travel model exist. It is also important to note that study on long-distance travel model usually face with lack of data. The challenge is that long distance travel is a rare cases part of daily mobility and thus experiences a very limited data availability (Frei, 2008; Moeckel et al., 2015).

#### 2.4.1 Heuristically derived parameters

Unknown parameters  $\beta_1...\beta_k$  of observable variables are often econometrically estimated from a sample of observation. Given the data-rich environment, parameters estimation can be done with ease; however, in the case of very limited data availability, parameter estimation would be a great challenging task. This is to say, insufficient data can yield weak coefficients (Moeckel et al., 2015), and that is considered implausible as well. Therefore, in light of theses stressors, some scholars have turned to the application of heuristically derived parameters (Moeckel et al., 2010; Alliance Transportation Group, 2015). The value in deriving a model is obtainable through a number of methods (i.e., previous mode choice modeling experience, literature review, adjustment of parameter, and sensitivity analysis) (Moeckel et al., 2015). One favored study with derived parameters for this thesis can be referenced to a previous long distance mode choice study conducted by Moeckel et al. (2015). The decision for it could be explained by the transferbility of the coefficients derived in that paper.

Moeckel et al. (2015) conducted a study on long distance travel model in North Carolina. The study used the derived parameters to formulate mode choice model rather than estimating model parameters. They derived the parameters based on their accumulated experiences and through literature review. A couple of tasks were exercised, including comparing parameters with other models and conducting series of sensitivity tests to ensure the reasonability of parameters selected. Finally, adjustment of the parameters was carried out to explore the impact of single parameters. The derived parameter results are found to be rational within the range of other models reviewed in the paper.

#### 2.4.2 Remark

The models reviewed in Section 2.4 were originally described and discussed in the study by Moeckel, Fussell, and Donnelly (2015). This thesis only re-discusses the models based on the original works.

#### 2.5 Route Assignment

The last step of four-steps model describes the allocation of the traffic flows on the transport infrastructure. This step fundamentally calculates the routing of the road users, and consequently the load on the routes and nodes on the basis of an integrated network model.

It is to be noted that there are three important information required for solving and formulating a traffic assignment model. First information is travel demand which is systematically done in the three earlier stages of the conventional four-steps model. Second required component is the characteristics of transport supply. The model needs to determine the correlation between travel demand and the transport system. The last information is a method of evaluating the corresponding distribution of the travel demand over the transport system (IHT, 1997, p. 91; Chow, 2007). All-or-Nothing, Wardrop's equilibrium, incremental, or capacity-restraint assignment are the most common approaches used for the route assignment as details in the following sections.

#### 2.5.1 All-or-Nothing

All-or-Nothing (AON) is the simplest model vis-à-vis other assignment methods. AON is based on the assumption that the road networks are free of congestion and the volume of traffic does not affect the travel time and cost. That is to say that travel time and cost are fixed inputs and independent from congestion effects. This model exhibits unrealistic situation as all the drivers perceive the same travel time and cost and subsequently choose the same route even there is another path with nearly the same travel time or cost (Mathew & Rao, 2007, p. 2). However, this model may be deemed reasonable in the case of sparse and uncongested networks where few alternative routes and a large variation in travel cost present. Virtually, the most important application of AON in practice is being a building block for other assignment methods.

#### 2.5.2 Capacity-restraint Assignment

Capacity-restraint assignment aims to approximate an equilibrium solution by involving a sequential application of iterated All-or-Nothing assignment. The function then uses congest function to recalculates the link travel times in relation to link capacity (Victor & Ponnuswamy, 2012). Conventionally, route choice has been calculated on the basis of ideal travel times in an empty network. The link travel time used in assignment is a function of speed on that link. That is to say that travel times will increase in a congested network when speeds decrease. The CR-Function is modeled as follows:

$$t = t_0 \left(1 + a \left(\frac{V}{C}\right)^b\right) \tag{2.6}$$

(Equation source: US Bureau of Public Roads, 1964)

where:

t : congested travel time

- $t_0$ : travel time (free flow)
- V : traffic volume on link
- C : capacity of link

a and b : parameters



Figure 2.2: Graph of CR-Function with different parameter of b Source: US Bureau of Public Roads, 1964

#### 2.5.3 Incremental Assignment

Incremental assignment is mathematical process in which traffic volumes are incrementally separated into a number of layers that are successively assigned to the best paths with the All-or-Nothing method. Traditionally, three iteration steps are allocated with a fixed proportion of travel demand (e.g. 50%, 30%, and 20%). Travel times on the link are then determined according to the traffic volumes on the link in each iteration step. When there are many increments used in this method, a set of possible errors in evaluation process may occur because of the discrepancies between travel times and traffic volumes on link. Also, incremental assignment often produces additional systematic biased results due to the influence of the assignment's order of OD pairs' volumes (Mathew & Rao, 2007, p. 6).

#### 2.5.4 Wardrop's First Principle - User Equilibriums

User equilibrium refers to as Wardrop's first principle or Wardrop's equilibrium. The principle is based on the assumption that every road user tries to minimize his or her individual travel time (or, more generally, to maximize its benefits). The resulting state of equilibrium denotes that no road user can further reduce his/her effort. Wardrop's first principle describes the user equilibrium in choosing the route alternatives r for each element  $F_{ijk}(t) = F_m$  of the trip matrix. The formula is expressed as follow:

$$T_{r}^{m} = \begin{cases} T_{min}^{m} & \text{if } F_{m,r} = F_{m}w_{r} \\ > T_{min}^{m} & \text{if } F_{m,r} = w_{r} = 0 \end{cases}$$
(2.7)

(Equation source: Treiber, 2017)

where:

 $T_r$ : travel time (generally: negative utility) of route alternatives r

 $T_{min}$ : minimum travel time

 $w_r$ : the proportion of the travel matrix element which has been allocated to the route r

Wadrop's first principle remains questionable as some analysts have probed whether the Wardrop's user equilibrium in practice (at least in a good approximation) is attainable. The assumption of individual benefit optimization is very plausible. However, the assumption of complete information and an objective definition of the benefit for all road users seems questionable (Treiber, 2017).

#### 2.5.5 Wardrop's First Principle - Stochastic User Equilibrium

User equilibrium assignment is a deterministic equilibrium which roots in an assumption that travel costs on used routes are equal and less than those on unused routes. Also, Wardrop stated that such model assumes the road users to have a perfect knowledge about the traffic condition (Gupta, 2010). Unlike user equilibrium, stochastic user equilibrium allows minimum (perceived) travel cost to be modeled for traffic assignment. The underlying assumption is road users perceive the travel cost on any given route differently and that the routes with the most minimum (cheapest) travel cost attract most trips between each OD pair (Mathew & Rao, 2007, p. 6). The models have distinct advantage over other models owing to their stability and independence of flows and they are less sensitive in response to slight changes in network definitions or link costs. The good aspect of the model is that they take multiple routes into account at one trip assignment and assign the trips on account of their good characteristics (Gupta, 2010). The models are practically most suitable for use in uncongested traffic conditions (Mathew & Rao, 2007, p. 6).

#### 2.6 Traffic Simulation Application

Transport modeling software plays a critical role when analyzing the travel demand. To facilitate the study, traffic analysts often use computer-based traffic simulation that can model the entire traffic network as realistically as possible. These software has been a realistic and useful tool for researchers and expert to realize the effects of any transport related projects in advance and to optimize individual components. Moreover, many different scenarios can be simulated and evaluated in a short period of time and with relatively little personnel and financial outlay.

#### 2.6.1 Overview of VISUM 17

VISUM is macroscopic traffic simulation software developed by PTV AG in Karlsruhe, Germany. The software is often used by many traffic analyst around the world for prediction of the changes in traffic network at aggregated zonal level. Four-steps model, the conventional macroscopic traffic model, is also available in the software.

Traffic engineers and transport planners use VISUM not only for transport planning, but also use it as a powerful analysis and data management tool. What is good in VISUM is its ability to detail the sophisticated routes and schedules of public transport service with a data model that goes far beyond traditional demand models (PTV VISUM Manual, 2017). Apart from the software functionality, the transport network in VISUM consists of links, nodes, transit routes, stops and so-called centroid connectors, which link all the zones with the transport network. Characteristics of vehicles such as size, type and weight are not considered in VISUM. Average speed, flow and density are rather used to dictate traffic condition. This is of particular importance in relation to macro simulation.

#### 2.6.2 Mode Choice in VISUM

VISUM offers variety of approaches for mode choice analysis. With the traditional four-steps model, traffic planners can use either Logit, Probit, Kirchoff, or Boxcox to determine the utility function for choice sets. As for logit model, the program is able to apply either binary logit or multinomial logit to determine the probability with parameters estimated from empirical data. The same approach also applies to probit model, either with binary or multinomial choice. Logit model is chosen for this study because it is deemed suitable due to its simplified approach with

easy estimation and interpretation, and it is applicable for transit strategies (NCHRP Report 365, 1998; Koppelman & Bhat, 2006). To facilitate the analysis, measurable unit or impedance of each alternative are required to determine what mode will be used. The input of impedance is generally expressed in the form of travel distance, and relative travel time. VISUM calculates this mode-specific impedance between all zones and later summarizes into matrix table called skim matrix. As for private transport, VISUM also allows users to model cost-specific attributes (e.g., travel cost, toll fee) for the study. On public transport side, timetable approach can be used to calculate the impedance for the public transport. With timetable approach, the program is able to compute all required variables for the in-vehicle-time and out-of-vehicle time (e.g., walking time, waiting time, and transfer time).

#### 2.6.3 Route Assignment in VISUM

#### Private Transport Assignment

VISUM offers users a wide range of assignment methods for private transport assignment procedure. One of the assignment procedures chosen for the study is user equilibrium called the equilibrium assignment in VISUM. As previously mentioned in Chapter 2, Section 2.5.4, the assignment is popularly used by many modelers. The assignment used Wardrop's first principle to distribute the demand. Based on a incremental assignment as an initial solution, the equilibrium state is calculated in a multistage iteration. This assignment method divides demand proportionally over the number of iteration steps defined by a user (Piatkowski & Maciejewski, 2013). The system then begins by checking if the routes with lower impedance can be found in the network. This raises a question on why the system has to check the alternatives with low impedance. Basic principle is vehicles will be loaded on the route with lower impedance; however, they will be shifted to a new route if another new alternative with lower impedance is found for a given connection. The load is distributed to the corresponding routes until all routes have the same impedance. The balance of the load is performed in pairs. That means the route with the lowest impedance receives a higher load than the route with the highest impedance. If no new routes are found, the iteration stops and the user balance is reached (Wermuth, Sommer, & Wulff, 2006).

#### Public Transport Assignment

As for public transport assignment, VISUM offers three possible search methods (i.e., timetable, headway, and transport system-based) for the analysis. The transport system-based, the simplest PuT assignment method, considers only the associated travel time of the network, and does not distinguish between individual PuT lines (PTV VISUM Manual, 2017). For the head-way based, each line route is described and run times between line stops and headway are taken into account. However, this procedure does not consider the timetable coordination. In this thesis, timetable-based is chosen for the study as it yields better results compared to other two methods. Timetable-based works on the assumption that passengers are aware of the available PuT timetable and use it to find the shortest path between designated OD pairs. Timetable-based involves the precise departure and arrival time of all services of public transport network as a guidance in the path search procedure (PTV VISUM Manual, 2017). In addition to the timetable, this method also considers the coordination of the timetable to help ensure precise results.

### Chapter 3

# Methodology

The framework of the proposed methodology in the study is presented in Figure 3.1 below. The methodology is structured into three main components: Data Collection, Data Analysis and Processing, and Model Building in VISUM.

To help facilitate the study and reach the objective of the research, a few set of data are required, such as commuting data in Munich Region, passenger survey to the Munich Airport, transit schedule, and geospatial data. Departure-arrival data and commuting data in Munich Metropolitan Region (MMR) are provided by the Assistant Professorship of Modeling Spatial Mobility (MSM). Passenger survey data to Munich Airport is provided by Flughafen München GmbH (FMG). Apart from travel data, geospatial data required for VISUM can be downloaded from Openstreetmap, a platform for freely-reusable geospatial data.

The next step would then be the analysis of the collected data. After the exploratory analysis stage, OD matrices are then generated based on the commuting data and passenger survey to airport. For transit schedule, General Transit Feed Specification (GTFS) is chosen due to its simplicity and its open-source feature. GTFS is created by using the departure-arrival data. Before the model in VISUM is built, the data should be aggregated by the influence zones which is delineated as a spatial unit. Irrelevant data to study area, such as data outside study area, are discarded. Model for the study is then created by using the filtered data. Model diagnostics is carried out to improve the built model. Finally, the model can be used to assess and forecast the travel demand.

Figure 3.2 presents the proposed four-steps model in this thesis. As previously mentioned above, OD matrices are generated based on commuting data in MMR. By adopting the OD matrices, the study saves the analyst time to construct a OD from scratch (through trip generation and distribution stage). This project focuses only on home-based trip, thus three activity pairs are created for the study, namely trip to work, trip to others, and trip to airport. The demand strata are therefore generated according to the activity pairs and person groups (HW, HO, and HA). Demand strata for the return trip (WH, OH, and AH) are not included in the study. A brief explanation on this matter is given in Chapter 4: Dataset. In mode choice stage, trips of each demand stratum are then separated for two particular transport modes, private and public transport. Finally, trips are assigned along the road network in the last step (Route Assignment). Once the base model is firmly built, the analysis is then proceed to the comparison between base scenario and the scenario with Erding Ring Closure.

#### CHAPTER 3. METHODOLOGY



Figure 3.1: Proposed Methodology Framework



Figure 3.2: Proposed Four-steps model for the study

### Chapter 4

# **Data Analysis and Processing**

This chapter explores the method to assess and process the dataset for the study. The first section begins with the method to generate OD matrices from the dataset. Then it outlines the incorporation of exogenous data into the OD matrices. Furthermore, it discusses the assessment of OD matrix. Finally, it briefly explains how GTFS is created and the geospatial data is cleaned through OSMOSIS.

#### 4.1 Generating OD Matrices

The current work relies on two main dataset to formulate the OD matrices for the study. First dataset was provided by the Assistant Professorship of Modeling Spatial Mobility (2017) and retrieved in the form of a CSV file. The dataset are already categorized for four modes (i.e., auto, transit, cycle, and walk). Each dataset contains the Origin and Destination Points for all trips recorded. There are only four trip purposes, namely home-based work, home-based other, and the other two return trips.

As a first step, data with bike and walking were discarded as they were not considered in the study. Then, data with auto and transit were separated by trip purpose which resulted in 8 dataset, 2 for home-based work, 2 for home-based other, 2 for work-home, and 2 for other-home of auto and transit respectively. Secondly, each dataset was then geographically filtered out by the zone of influence which is delineated as spatial unit. Since the model is being built for Munich region, the only trips that were considered were those originate and terminate within Munich region (see Chapter 5 for study area). Trips outside the study area were later removed. This resulted in 2,796 OD pairs for each trip purpose. Table 4.1 summarizes the number of trip by purpose.

Mode	Activity Pair	Description	Total
	НО	Home-based other	618,285
Auto	HW	Home-based work	$365,\!457$
Auto	OH	Other-home	$618,\!285$
	OW	Work-home	$365,\!457$
	НО	Home-based other	238,370
Transit	HW	Home-based work	$138,\!134$
Transit	OH	Other-home	$238,\!370$
	OW	Work-home	$138,\!134$

Table 4.1: Summary of trips by purpose

The number of OH and WH trips shown in the Table 4.1 are found to be equal to HO and HW. This can be logically explained by the assumption that return trip should be equal to outbound

trip. According to the dataset, trips with travel distance more than 80 km were discarded. Trips recorded in the dataset are based on trip distance, and thus does not fit to represent mode share. It is worth noting that the dataset contain only the Home-based trip, not the Non-Home-based trip. This is because the data for Non-Home-based trip are not available.

Second dataset used to generate OD matrix for airport is based on passenger survey data provided by Flughafen München GmbH (FMG). Dataset, which is surveyed in 2016, describes the person trip between two modes (i.e., auto and transit). Trips were directly tabled into matrix by assuming one destination for all. For example, each trip is terminated at airport destination. Trip to airport were classified as homebased-airport (HA). Commuting trip to airport is not included in this dataset as it was already covered in HW trip.

#### 4.1.1 Incorporating exogenous data

Commuting data in MMR contains all the information of each trip with certain origin and destination. OD matrices generated from it can be applied and used directly as a distribution model. However, the data is limited only to home-based work trip and home-based other trip. A shortcoming of the data is the lack of commuting data to Munich Airport. Two major tasks were involved to incorporate the passenger data into the study. First task focused on the distribution of trips to its originating zones whereas second task attempted to allocate the trips to its destination zone.

The trip origins recorded in the survey were coded to the system of districts used by FMG. However, at this level of geographical resolution, the resulting data is not considered convenient to represent the trips in each TAZ for airport bound trip. For the purpose of model estimation, the surveyed annual passenger trips to airport was aggregated to a system of TAZs instead of districts resolution. Since the given data was recorded in annual number, an assumption was made to create trip person end (trip per day). It was assumed that annual trips were made in a common calendar year of 365 days. The total number of trip to airport was the sum of total trips per day from each district. Therefore, trip origins from each district were divided by 365 days. Then, the trips were distributed to each TAZ by scaling with the proportion of population in each TAZ.

After trips were distributed to its originating zones, trips were then allocated to four zones that located adjacent with the airport area. Distribution of trips at destination zones were undertaken following the proportion of commuting trip - HW. Airport lies across four zones, and the zone with highest incoming HW trip will receive higher number of airport trips as well. Moreover, special condition was set if all four zones have zero incoming HW trip, airport trip is allocated to zone number 245. The reason for it is most companies' offices locate in zone number 245.

#### 4.1.2 Assessing OD Matrices

Each trip has principally two ends, trips originating in a zone (production end), and trips are destined for a zone (attraction end). Theoretically, the total number of trips produced per zone should be equal to the total number of trips attracted per zone. In practice, however, the estimation of these two trip-ends will not be equal. Under this assumption, balancing method is usually applied at the end of trip generation stage in order to make production equal to attraction trips or vice versa. In particular, the selection of which trip-end to be balanced is done based on the degree of confidence. Trip production has normally greater degree of confidence since it is based on household travel survey, whereas the attractions are based on characteristic of data (location of study area and some external information) (Skarphedinsson, 2009, p. 22). Balancing the producing and attracting trips for each trip purpose is done individually. Between production and attraction trips, if one of them is used to control the

total, the other is no longer relevant for the study.

Selection of which trip to be modeled can be referenced back to the balancing approach reviewed above. HW, HO, and HA will be used to to formulate and analyze mode and route choice in VISUM. The return trips WH, OH, Airport-Home (AH) will not be used. By virtue of balancing method, the model can distribute trips done in the direction towards the attractions with higher degree of confidence than towards the productions.

#### 4.2 General Transit Feed Specification (GTFS)

Initially, I considered using public transport timetable from Münchner Verkehrs- und Tarifverbund (MVV) and Stadtwerke Landshut to manually create the transit schedule. However, it was not feasible to do it as we have more than 500 transit line routes within the proposed study area (see Chapter 5). Therefore, an option with GTFS was chosen.

General Transit Feed Specification (GTFS) was created and provided by Dr. Carlos Llorca Garcia from the Assistant Professorship of Modeling Spatial Mobility at Technical University of Munich. Data used to formulate GTFS was retrieved in the form of CSV file. The given data recorded most of public transport stop points in MMR. Due to the different patterns of each line routes, the following assumptions were made to facilitate the study process:

- 5 minutes for U-Bahn
- 10 minutes for tram and bus in core city
- 15 minutes headway for bus in small city
- 20 minutes headway for S-Bahn and bus in rural city
- 30 minutes for regional train
- Operation time from 4h00 to 22h00 without peak or off-peak time
- Daily Schedule

It is true that some assumptions do not reflect the actual schedule. For instance, U-Bahn does not run with a constant headway of 5 minutes from 4h00 to 22h00. However, for convenience, it was decided to assume constant headway for all line routes. Moreover, there are 1,398 line routes in MMR created for GTFS. Considering a proper schedule of each line route, the creation task would be very time-consuming as each line route has its own specific number of vehicle journeys, different headway for morning peak and evening peak, and different schedule for weekdays and holiday. Besides, data used to create GTFS was generated through Google API, and thus some data for regional bus was not accurate either. For the instruction on how to create GTFS, the information can be referred to: https://developers.google.com/transit/gtfs/reference/

#### 4.3 Geospatial data

Geospatial data used in VISUM was downloaded from Geofabrik, a geodata distributor of Openstreetmap. Before the data can be used to model the travel demand, some networks had to be cleaned to fit the model resolution. The resolution of the study is at the district level. Hence, the residential and unclassified streets are not considered in the model. However, it was later decided to include residential and unclassified streets in the model as the analysis proved that many disturbances on transit line routes have been occurred. An example for it is the line routes of bus has to be bundled in the other road levels (e.g., primary and secondary). To clean the network, a few tasks were undertaken through OSMOSIS, a command line Java application for processing OSM data. Firstly, the downloaded geodata was geographically cut using the polygon command against the boundary of study area. Secondly, some unwanted links and nodes were discarded by using the pipeline control task. An example of command used to assess OSM data is illustrated in following Figure 4.1.



Figure 4.1: OSM Data Manipulation Task Source: OSMOSIS Detailed Usage 0.46

### Chapter 5

# Study Area

Munich is the capital city of German state of Bavaria. The city is the third largest city with high density of inhabitants after Berlin and Hamburg. Munich is well-known for its automobile industry as a city itself is a major economic center in southern Germany. In terms of metropolitan status, Munich Metropolitan Region (MMR), one of the eleven metropolitan regions in Germany, principally consists of Munich and its surroundings. MMR is currently a home to more than 5.2 million people (Metropolregion München, 2015). In 2016, it was reported that more than 368,000 commuters traveled daily to work in Munich from its surroundings (Planungsverband Äußerer Wirtschaftsraum München [PV], 2016). To avoid any confusion between Munich and the district of Munich, I decided to referred the capital city as Munich City and district of Munich.

#### 5.1 Selection of study area

This section explains how study area in the research is determined. As previously discussed in Chapter 1, Section 1.3, the objective of the thesis is to investigate any possible impacts of Erding Ring Closure on passenger flows. It is worth noting that the goal of Erding Ring Closure Project is to improve the airport access as well as the transit connection in Erding. Therefore, the main criteria in defining the planning area for this thesis are limited to the following criteria:

- Data availability
- Commuter Data to Munich City, Freising, and Erding
- Travel survey of commuter to Munich Airport
- Location of Erding Ring Closure, S-Bahn S1, and S-Bahn S8
- Socioeconomic and demographic data
- Other constraints (e.g., time constraint for the whole research)



Figure 5.1: Study area for this research Source: Openstreetmap and elaborated by author with ArcGIS

Among many potential locations, the choice of two urban and five districts (as shown in Figure 5.1) for the study was influenced by their high share of commuting trips to Munich City, Erding, and Freising. Another more plausible account of the reason to select them is the trips to airport. To elaborate upon the main criteria, I have made an assumption regarding the number of commuters to Munich City. A quick analysis is carried out to support select the data relevant to study's purpose. It should also be noted that the selection process is limited to the available data provided by the Assistant Professorship of Modeling Spatial Mobility (MSM).

At first step, districts with the number of daily commuters to Munich City lower than 10,000 will be discarded. Data generated by Federal Agency for Employment (German: *Bundesagentur für Arbeit [BA]*) in 2016 recorded the districts with daily commuter with social insurance to Munich City. Based on the statistics, only 7 out of 27 available urban and districts has the number of commuters more than 10,000. These 7 districts are Dachau, Ebersberg, Erding, Fresing, Fürstenfeldbruck, Munich, and Starnberg.

The selection of study area then continued based on the assumption that if there is no significant changes on current travel times from any districts to airport or changes on passenger flows at certain station, that districts will be excluded. Dachau, Fürstenfeldbruck, Starnberg are connected to airport by S-Bahn S1 and S8. By taking travel time as a main indicator, it is
obvious that commuters from Dachau, Fürstenfeldbruck, and Starnberg would remain taking S-Bahn S1 or S8 to the airport instead of the new service. The addition of new service to airport lines would not make any changes to the current travel time for them. Importantly, the distribution of passenger loads on 2nd Main Trunk (German: *2.Stammstrecke*) would not change either since the situation remains almost the same. Ebersberg, on the other hand, will now have two choices at the Munich's east station (German: *Ostbahnhof*). Commuters can either take S-Bahn S8 or S2 to airport. Travel time on S-Bahn S2 is perhaps longer by a small margin; however, the passenger flows at Ostbahnhof is evidently like to be influenced by the new service. Therefore, Ebersberg is included in the study area while Dachau, Fürstenfeldbruck, and Starnberg are excluded.

Landshut City and Landshut are later included due to their significant share in passenger survey to airport, as well as its underlying status in the Erding Ring Closure project. The inclusion of Landshut can be partially explained in the context of travel time of Landshut-Messe. Under this context, it is possible that Messe bound commuters are like to go through Erding rather than to travel through Munich's central station (German: *Hauptbahnhof*) when the extended line is put into operation. Apart from the Landshut-Messe situation, the data provided by FMG was also used to help better the decision on Landshut. A survey done by FMG in 2016 showed that the share of passengers from Landshut is significant higher compared to other districts in the region. Thus, it makes Landshut City and Landshut applicable for the study area.

## 5.2 Current situation of urban mobility

This section briefly describes the current situation of urban transport and mobility in the study area. However, there is no available data for Landshut City and Landshut. Therefore, only 5 urban and districts are discussed in the context of urban mobility. Moreover, the average value is used to explain the current situation instead of using each value from each district. Those values can be found in car ownership chart, car availability chart, and accessibility to workplace chart.

Data generated by the Mobility in Germany (German: *Mobilität in Deutschland [MiD]*) in 2008 found that the districts within the study area have higher rates of car dependency than average, with longer commuting distances and a spread out urban structure (MVV, 2010a,b,c,d,e). Using data from MiD for a mode share analysis of the districts in study area, the report provides an interesting insight into the share of travelers using a particular mode of transport, as shown in Figure 5.2 below.



Figure 5.2: Modal Split in districts within study area Source: Mobility Reports from various MVV Reports, 2010a,b,c,d,e

Figure 5.2 shows the modal split in 2008, and by far private car is the most frequently used means of transport. With the exception of Munich City, it appears that the shares of private vehicle from other districts are higher than 50% (sum of auto-driver and auto-passenger). According to MVV (2010), the report highlighted that walking and cycling have important roles in daily mobility, however most of this is concentrated in the city.



Figure 5.3: Number of car per household Source: Mobility Reports from various MVV Reports, 2010b,c,d,e



Figure 5.4: Car availability as a driver Source: Mobility Reports from various MVV Reports, 2010b,c,d,e

Figure 5.3 and Figure 5.4 provide additional information on modal split within the study area. In average, the car ownership bar chart shows that only 6.5% households do not possess a car compared to 93.5% households with cars, in which 48.5% households having access to one car. In other words, every household in study area has an average of 1.4 cars (MVV, 2010b,c,d,e). Notably, the high car ownership also proves that commuting with private car is the most popular choice within the city, and only those who possess no driver license cannot access a car. This can partially be explained by the sparser public transport coverage outside the city, the reduced flexibility buses offer, and the fact that cycling and walking are not competitive with the comfort provided by a private car, especially over longer distances or in bad weather.



Figure 5.5: Accessibility to workplace with transit Source: Mobility Reports from various MVV Reports, 2010b,c,d,e

In contrast to the other transport modes, it appears that public transport in study area occupies only 8.75% in mode share (average value with exception of Munich City) based on Figure 5.2, and this number is even lower than both walking and cycling if combined. The fact that public transport is not so widely used within the city provides an initial clue that there may be poor public transit accessibility or service in these areas. It is likely that individuals use private car over the public transport for short-distance trips. As further proof of this, Figure 5.5 provides additional information regarding residents' accessibility to their workplaces by public transport. It shows that 55.75% of commuters rated the accessibility to their workplaces by public transport as bad, very bad, or as not at all. According to the interviews conducted by MVV, commuters would like to see some significant improvements of public transport (MVV, 2010b,c,d,e). Should there be an improvement, commuters may swap their private car trips for public transport trips in the city, especially where distances are short.

### 5.2.1 Road Network

The road network in the study area comprises of five autobahn (i.e., A92, A93, A94, A99, and A8). As of 2016, autobahn network in Germany has a total length of about 12,993 km, in which 2,515 km is in Bavaria (Statistische Ämter des Bundes und der Länder, 2016). Most sections of autobahns contain two, three, or sometimes four lanes with an additional of emergency lane. There are also some other sections that remain in an old state, with two lanes and no emergency lane.



Figure 5.6: Private network within study area Source: Openstreetmap and elaborated by author with ArcGIS

From Figure 5.6, Munich City has the best connection compared to other districts in study area. The ring road, A99 provides a better access for car to other motorways. Ebersberg and Erding are the only two districts that are not connected by autobahn. Indeed, these two districts are connected by federal highway B301 and B388 respectively. Besides, it seems that access to airport by road network is well-established. Airport can be reached by A92 and B301 on the western corridor while a number of local road networks on the the eastern side.

#### 5.2.2 Public Transport Network

Public transport in Germany is operated by different transit authorities, depending on where they locate. For instance, Münchner Verkehrsgesellschaft (MVG) is a municipally owned company that operates U-Bahn, tram, and city bus in Munich City. MVV, on the other hand, is a transit authority of Munich City. MVV does not provide any transport service, but coordinate transport and fare in and within surrounding areas of Munich City. Transport services are indeed provided by 40 companies under the supervision of MVV (MVV, 2015). For example, S-Bahn is solely operated by S-Bahn München, a subsidiary of Deutsche Bahn (DB). In Landshut, Stadtwerke Landshut is the main transit authority of the city that operates city bus, express bus, and airport express. On regional level, DB provides all the service of Regio Bahn throughout the whole country.



Figure 5.7: Public Transport in districts within study area Source: Openstreetmap and elaborated by author with ArcGIS

Based on Figure 5.7, Munich City and Munich have the most well-integrated public transport network with most stations are multimodal stations. In terms of rapid transit connection, a total of 8 S-Bahn lines are in operation with 8 branches on western side and 5 on the eastern side. Today, the S-Bahn covers most of the populated area of the Munich Metropolitan Region, connecting most cities in surrounding districts. The airport is connected to the capital city by S-Bahn S1 and S8. The journey to airport from Erding with S-Bahn is not possible as the last stop of S-Bahn S2 terminated in Erding. However, the city is connected to airport by regional bus which operates with 20 minutes headway, depending on the peak time. The connection between Landshut and airport is bridged by airport express line and regional train service with additional transfer in Freising.

## 5.3 Analysis of travel times

A discussion of relative travel time is made in addition to the current situation of mobility in study area. It may be worth noting that the Erding and Landshut have the most poor public transport connection compared to other districts in the study area. Performing the travel time analysis resulted in contour maps of both isochrones.

While these isochrones can be used to make judgements about the temporal accessibility via a certain mode, they are not as helpful for identifying areas where public transport are uncompetitive with private car. For this analysis, mapping the relative travel time of public transport against that by car is more helpful to visualize areas that require improvement. Figures 5.8 and Figure 5.9 show this relative travel time by public transport and car respectively.



Figure 5.8: Transit travel time to airport in minutes Source: Google Maps and elaborated by author with ArcGIS

Figure 5.8 presents the travel time with public transport to the airport. It appears that almost entire area of Landshut have no transit access or poor access to airport, except for Landshut City and other small towns along the rail corridor. The journey to airport from Ebersberg and Erding would take between 75 to 169 minutes (based on Google Maps) which is considerably high, twice the amount of travel time from Munich City to airport. The figure above also highlights the short travel time along S-Bahn S1 and S8 lines to airport. Those who live anywhere close to S1 and S8 lines can reach airport within or less than an hour.



Figure 5.9: Car travel time to airport in minutes Source: Google Maps and elaborated by author with ArcGIS

While public transport is more competitive in the city center and southern half of Munich, it is broadly uncompetitive in the northeast of Munich, with the worst areas lying from Erding to Landshut. In analyzing the Figure 5.9, it is generally said that private car is more competitive than public transport. While the journey to airport with public transport takes at least 79 minutes, travelers from Erding and Landshut can reach the airport within 30 minutes by car. In contrast to public transport, it seems that travelers from Landshut can access the airport with private car regardless of where they live.

## 5.4 Transport Analysis Zone System

Defining the zone system is an essential element of the transport modeling framework. The zone system in rural area is designed to be larger than the zone in urban area. The TAZ is created by MSM through the rasterization method by Moeckel and Donnelly (2015). The zones for study area are made up of 2,796 zones across five districts and two urban. Both Figure 5.10 and Figure 5.11 show the zone of the study area.



Figure 5.10: Traffic zones of study area within MMR Source: MSM and elaborated by author with ArcGIS



Figure 5.11: Traffic zones of study area Source: MSM and elaborated by author with ArcGIS

### 5.4.1 Airport Zone System

Due to the predefined raster cells, four raster cells are found to be adjacent with airport area. Therefore, to apply and analyze the scenarios in this area, four zones surrounding the airport are modeled as airport zone.



Figure 5.12: Zone system of airport Source: MSM and elaborated by author with VISUM

# Chapter 6

# **Modeling Framework**

#### 6.1 Model Specification

The specification of the model used for this study is described in this section before the estimation process. This section involves specifying the choice set, explanatory variables, and model structure. As stated in Chapter 1, the choice set was decided to be auto and transit.

#### 6.1.1 Variables considered

As described in Section 2.2.3, the traveler's preference of alternative choice sets can be predicted by adopting the utilities of the alternatives, and these utilities are in turn governed by various explanatory variables. Hence, it is necessary to determine which variables should be included in the design once the alternatives has been decided upon. According to Ortúzar & Willumsen (2011), the variables used to determine the choice of transport mode are normally based on socioeconomic factors, trip characteristics, and transport facility characteristics. Although it may have been possible to include all these variables, it was decided to use only specific variables related to transport facility. This includes specific travel time and cost variables. However, it would be difficult to create common variables for most of the variables included. Main reason can be referenced to a number of the attributes that only pertained to one alternative. An example of which is the number of transfers in public transport. This attribute is obviously only relevant for explaining the public transport and not the car. Finally, to make the model more manageable, efficient and balanced, only five variables are assumed for each of the alternatives. Some variables included are adopted from previous studies that yielded results with high validity while the specification of some variables are created based on a couple of assumptions.

#### Auto access and egress time

The reason to create access and egress time as a separate variable was based on several pieces of information. Moeckel et al. (2015) noted that the coefficient of out-of-vehicle time is twice as negative as the coefficient of in-vehicle time. This is mainly because the times spent outside vehicle were perceived as burdensome than the travel time in vehicle. As for auto, access time refers to the time spent to access the vehicle, and this also includes the walk time to the garage or parking lot. Access time is assumed to be 2 minutes in this study. Clearly, this value is quite low for some areas where travelers walk from home to the parking lot to access the vehicle. In most cases, however, 2 minutes is deemed reasonable as travelers usually park their own vehicle right in the garage or in front of the house. On the other hand, egress time refers to the time spent to walk from the parking location to the final destination. Egress time varies greatly according to the location of the final destination. In urban areas, egress time is somewhat high compared to the suburban and rural areas where travelers can park right in front of the destination (Moecket et al., 2015). In this study, 5 minutes is chosen for the variable, which is fairly high but reasonable. The attribute used was based on those used in the mode choice study conducted by Moeckel et al. (2015).

#### Travel time by car

Travel time generally represents the level of service (LOS) of transport mode in the network. In this study, the absolute fixed value of travel time by car could not be used due to the wide range of travel times included in the design. Indeed, the travel times by car were estimated by using VISUM. To estimate the travel times, first the generated OD matrices (previously described in Chapter 4) were allocated into its TAZs. Then VISUM calculated the travel times and depicted it in the form of skim matrix. Using Google Map to engineer the travel time is also possible; however, it was decided not to use it due to the extremely time-consuming process.

#### Auto operating cost

Operating cost normally covers gas price, maintenance, insurance, tax, and other costs associated with car usage. It was decided to use operating cost variable, which includes the average fuel consumption cost per kilometer, fixed costs (insurance and tax), and variable costs (maintenance). The fixed value used as the reference was adopted from the report prepared by the Allgemeiner Deutscher Automobil-Club e.V. (ADAC) (2017). The report provides the average all inclusive driving cost of over 1900 vehicle models. For this study, the value of operating cost was 0.47 EUR/km, which was the median value.

#### Parking cost

In contrast to operating cost, it is difficult to define metered parking fee as it differs considerably depending on the location. The parking fee was initially modeled based on the parking rate in Munich's city center, which is 0.50 EUR per 12 minutes. Considering the case of commuting trip, if the vehicle were at the parking place for 8 hours, the resulting parking fee would be 20.00 EUR. Clearly, this makes the parking fee more dominant in the model. Therefore, the parking fee rate of 0.50 EUR per 12 minutes was not used. To tackle these obstacles, it was decided to adopt the value of parking fee based on two parking studies in Germany and England conducted by Axhausen & Polak (1991). The fixed absolute value of parking fee are assumed to be 2.00 EUR for general parking places. In terms of parking fee at airport, it was decided to use 64.00 EUR which is the parking fee for two days. The reason for it is because air passenger tend to stay longer at the final destination. Hence, considering at least two days of parking seems to be convenient for the study. In order to be able to ensure the consistency, Moeckel et al. (2015) suggested to divide parking cost equally for outbound and return trip. This is because the travel time and cost are accounted for one way only.

#### Transit access and egress time

Similar to auto access and egress time, the transit access and egress time were created as separate variables from travel time. As for transit, access time refers to the time spent outside the vehicle, which includes the walk time to platform or gate. With regards to egress time, it refers to time spent from the platform to the final destination. In contrast to auto access and egress time, the value of access and egress time for transit were estimated with VISUM. Both times were calculated and represented by skim matrices in VISUM.

#### Travel time by transit

Generally, travel time by transit is measured with references to its components. This includes access time, in-vehicle time, origin wait time, transfer wait time, and egress time. However, since the access and egress time were separate variables, travel time could be measured in terms of origin wait time, transfer wait time, and in-vehicle time. To estimate the travel time by transit, the same approach used for access and egress time was applied. Firstly, all of its components - origin wait time, transfer wait time, in-vehicle time were calculated by VISUM. The results showed that transfer times were significantly higher than actual time (using Google Maps), whereas origin wait time was zero value for any given trip. This irrationalities could have been caused by GTFS data used to create transit schedule. A couple of assumptions were

made in GTFS, and some assumptions did not reflect the actual schedule. An example for it can be referenced to the one headway for both peak and off-peak time. In light of this obstacle, in-vehicle time was assumed as travel time by transit.

#### Number of transfer

Similar to access and egress time, VISUM generated the number of transfer according to the network pattern. Absolute value for this variable could not be used since the given trip were made for different destinations, thus makes it difficult to assume one fixed value.

#### Transit Fare

Along with travel time, the cost associated with a transportation mode, or transit fare, has major impact on mode choice (Ortúzar 2000; Vuchic 2005; Ahern & Tapley 2008; Twaddle, 2011). The transit fare for the public transport for the study was adopted from report prepared by Verband Deutscher Verkehrsunternehmen (VDV) (n.d.). The reference of the fare was calculated based on the rate of 2.10 EUR per 5.9 kilometer, and the resulting fare per kilometer is 0.35 EUR. The cross-check with MVV Tariff Structure for single journey trip was also attempted. The calculation showed similar results with minor difference.

#### Summary of variables considered

Table 6.1 provides a summary of variables that were discussed in previous sections.

Trip characteristics			
Mode	Variable	Specification	Source
	Access time	$2 \min^1$	Created for the experiment
	Egress time	$5 \min^1$	Moeckel et al. $(2015)$
Auto	Travel time	-	Generated by VISUM
	Operating cost	$0.47 \ \mathrm{EUR/km^2}$	ADAC (2017)
	Parking cost	$2 \text{ EUR}, 64 \text{ EUR}^3$	Axhausen & Polak (1991)
	Access time	-	Generated by VISUM
	Egress time	-	Generated by VISUM
Transit	Travel time	-	Generated by VISUM
	Number of transfer	-	Generated by VISUM
	Transit fare	$0.35~{ m EUR/km^4}$	VCD (n.d.)

Table 6.1: Variable considered in the estimation

<sup>1</sup> Fixed value

 $^{2}$  Average fuel consumption cost

<sup>3</sup> 2 EUR for general parking locations, 64 EUR for airport

 $^4$  Converted from 2.10 EUR/5.9km

#### 6.1.2 Utility equations

The preferred modeling framework is disaggregate discrete choice model. The main assumption is that trip-maker tries to maximize personal utility and selects the alternative with the highest utility. As described in Section 2.2.3, the utility function of two alternatives can be formulated as follows:

$$U_{Auto} = V_{Auto} + \varepsilon_{Auto}$$

$$U_{Transit} = V_{Transit} + \varepsilon_{Transit}$$
(6.1)

The corresponding deterministic part V of the alternatives expressed as the sum of the explanatory variables and unknown  $\beta$  parameters are formed as follows:

$$U_{Auto} = C + \beta_{OVT} * access time + \beta_{IVT} * travel time + \beta_{OVT} * egress time + \beta_{AOC} * travel distance * operating cost + \beta_{PC} * parking cost * 0.5 U_{Transit} = C + \beta_{OVT} * access time + \beta_{IVT} * travel time + \beta_{NTR} * number of transfers + \beta_{OVT} * egress time + \beta_{TRF} * travel distance * transit fare$$
(6.2)

#### **6.1.3** Unknown $\beta$ parameters

The review of literature with regards to the unknown parameters in Section 2.2.8 has provided a guideline to select model parameter in the case of very limited data availability. It was decided to assume the parameters from mode choice study conducted by Moeckel et al. (2015). It is worth saying that the parameters are heuristically derived parameters. The main decision to use these parameters was due to the lack of data availability in MiD. In particular, MiD offers a wide range of travel information in both urban and suburban areas for the whole Germany. However, MiD lacks specific data associated with travel time (i.e., access and egress time). This shortcoming thus makes MiD inapplicable for this study. Table 6.2, Table 6.3, and Table 6.4 summarizes the adopted parameters and constants.

Coefficient	Description	Personal	Commute	Source
$\beta_{OVT}$	Out-of-vehicle time coefficient for	-0.036	-0.050	Moeckel et
	auto: parameter evaluating the			al. $(2015)$
	travel time spend out of vehicle			
$\beta_{IVT}$	In-vehicle time coefficient for auto:	-0.018	-0.025	Moeckel et
	parameter evaluating the travel			al. $(2015)$
	time spend in a vehicle			
$\beta_{AOC}$	Auto operating costs coefficient	-0.008	-0.007	Moeckel et
				al. $(2015)$
$\beta_{PC}$	Parking costs coefficient	-0.012	-0.010	Moeckel et
				al. (2015)

Table 6.2: Coefficients for utilities	s of a	auto
---------------------------------------	--------	------

Table 0.3: Coefficients for utilities of transit	Table 6.3:	Coefficients	for utilities	of transit
--	------------	--------------	---------------	------------

Coefficient	Description	Personal	Commute	Source
$\beta_{OVT}$	Out-of-vehicle time coefficient for	-0.036	-0.050	Moeckel et
	transit: parameter evaluating the travel time spend out of vehicle			al. (2015)
$\beta_{IVT}$	In-vehicle time coefficient for transit: parameter evaluating the travel time spend in a vehicle	-0.018	-0.025	Moeckel et al. (2015)
$\beta_{NTR}$	Coefficient on number of transit transfers	-0.01	-0.01	Moeckel et al. (2015)
$\beta_{TRF}$	Transit fare coefficient	-0.012	-0.010	Moeckel et al. (2015)

The utilities are estimated against a base alternative. In the model, if one alternative was chosen as the base alternative, its constant has to be set to zero. It is important to note that the constants in Table 6.4 are used as a reference point. The calibration of the constants is later required to estimate the accuracy of model fit.

Constant	Description	Personal	Commute	Source
C	Constant Auto	$0.46 \text{ to } 0.58^1$	$-0.51$ to $-0.20^1$	Moeckel et
				al. $(2015)$
C	Constant Transit	0.52	0.30	Moeckel et
				al. (2015)

Table 6.4: Constant for auto and transi	Fable 6.4:	Constant	$\mathbf{for}$	auto	and	transi
---	------------	----------	----------------	------	-----	--------

<sup>1</sup> Distinguished by auto occupancy

## 6.2 Model Building in VISUM

A model in VISUM was developed by following the task described below. For this section, the reader is assumed to have a basic understanding of VISUM. Otherwise, the explanation of software can referred to the user manual. To obtain a better understanding of the software, this section is supplemented by the screenshots showing how VISUM was configured. This can be found in Appendix.

#### Initial setup

VISUM was opened, and a new project file was created. Selection of 4-Steps model was then confirmed in demand setting window.

#### Person Group

Mobility behavior of road users varies significantly from age to occupation and to other constraints (e.g., possession of driving license or vehicles). Basically, road users with similar mobility behavior are grouped together and broken down into socio-homogeneous group. For instance, person groups are commonly formed based on characteristics and occupations of road users. In this model, people who are employed can be grouped into a group called EMP and the rests are formed as POP.

#### Activity Pair

Trip purposes and activities in certain locations are pre-assumed to be the main cause for a particular movement. People move and travel to perform certain activities, and this explains the relation between activity and mobility. Work, School, Leisures are some of activity examples. Activity Pair is formed as a consequence of two successive activities in daily routine (PTV VISUM Manual, 2017). Three activity pairs were created, namely HO, HW, and HA (previously discussed in Chapter 3 and 4).

#### **Demand Strata**

Demand stratum links one or several person groups with activity. The combination of person groups and an activity forms a basic demand object which is mainly used in almost calculations of the first three steps. For example, the demand stratum "HW-EMP" represents the group of employees going to work.

#### **Demand Matrices**

Matrices were generated from the dataset (as described in Chapter 4). Matrices were imported to VISUM by using AddIn tools in Script setting window. Three demand matrices were created and named as Distribution HA, Distribution HO, and Distribution HW.

#### Private Transport Network Modeling

The next step to do after all required inputs have been settled is importing the cleaned network into VISUM. Link classifications had been predefined by Openstreetmaps based on German Road Classification. Thus, no extra task on link capacity and speed was exercised. VISUM offers four configurations of resolution to be analyzed, ranging from very detailed to a simple one. At first attempt, urban network setting was modeled. However, there was an error with bus line routes as it could not be bundled into the network, and hence false line routes were created instead. Later, it was decided to model detailed urban configuration, and the results has been improved though some false bus line routes still remain there.

#### Transit Timetable

GTFS was imported into VISUM through AddIn in Script setting window. Because a single line route requires certain streets and certain stop points, high standard deviation value was given to allow well-established allocation of line routes on certain links in target network. VISUM did not import only the line routes, but also the timetable that has been set up through GTFS. A total of 1,398 line routes with 87,334 vehicle journeys were obtained. A few modifications on timetable were also attempted in order to improve the accuracy of timetable before the analysis.

#### Zone System

Zone was modeled based on TAZ (as discussed in Chapter 5). Zone, which was retrieved in the form of Shape file, was imported into VISUM and labeled it as zone system. A total of 2,696 zones were made up for the study area. The size of the zone does not matter in VISUM as long as the centrioid are defined.

#### Connector

Connectors connect the active zone with transport nodes (e.g., stop point, or node on road). There are two types of connectors, private and public connector. As for private connector, maximum length of direct distance was set to 5 km. Public transport connectors was created based on the assumption that traveler has a fair access to stop area within the radius of 1.5 km. Thus, 1.5 km was assumed for public transport connector.

#### Summary of work done

Activity	Description	Type of trip	Person	Demand
pair		purpose	group	Stratum
НА	Trip from home to airport	Personal	POP	HA-POP
HO	Trip from home to other	Personal	POP	HO-POP
HW	Trip from home to work	Commute	EMP	HW-EMP

Table 6.5: Summary of inputs required to run model



Figure 6.1: Connector configuration in VISUM Source: Openstreetmap and elaborated by author with VISUM



Figure 6.2: Network modeling in VISUM Source: Openstreetmap, MSM and elaborated by author with VISUM



Figure 6.3: An example of network in city center Source: Openstreetmap, MSM and elaborated by author with VISUM



Figure 6.4: An example of network in airport area Source: Openstreetmap, MSM and elaborated by author with VISUM

# 6.3 Model Estimation and Results

#### 6.3.1 Estimating mode choice with VISUM

As stated in Section 2.6.2, it was decided to use logit model for this project. Logit model assumes that the error component is independently distributed and type-I extreme value (identically Gumbel distributed). The decision to use logit model was made based on its simple approach and straightforward specification. More importantly, the logit model may be the most suitable for this study because none of the alternatives are deemed to be significantly correlated.

#### 6.3.2 Inputs to VISUM

To simulate mode choice in VISUM, demand stratum of each trip purpose are required. Analysis was exercised in the case of binary choice, and thus trips of each demand stratum were separated between two transport modes (i.e., Auto and Transit). As Figure 6.5 illustrates, data used for the model step mode choice comprises skim matrices, utility function, and demand matrices of each trip purpose. Utility matrices are mainly calculated based on skim matrices.



Figure 6.5: Simplified flowchart of mode choice model in VISUM

#### 6.3.3 Simulation results

In this model, auto was chosen as the base alternative, with its constant set to zero. Adopting the constants from Table 6.4 as a reference point, the simulation results highlighted two different results. The derived coefficients appeared to be feasible for airport trip, with only small difference between predicted share and actual share. This could be partially explained by the characteristic of trip to airport, which is often modeled as long-distance travel trip. The parameters in turn was also intended for long-distance travel model. On the other hand, the presented results highlight high deviation of the mean mode shares predicted by the model compared to the actual shares for commuting and other trip. Figure 6.6 and Figure 6.7 compare the results between actual share and predicted share of auto and transit respectively.



Figure 6.6: Comparison between actual and predicted share of auto

As Figure 6.6 illustrates, the deviation of observed auto shares compared to the actual shares lies within a range of -2% to -30%. An example of it is the trip to other by auto whose predicted share is roughly 33% lower than the actual share.



Figure 6.7: Comparison between actual and predicted share of transit

As for transit shares, Figure 6.7 shows the distinct difference between the predicted share and actual share. The difference rates were found to be within a range of +2% to +32%. For example, VISUM predicted the mean mode share of trip to other by transit with 41.65%, which is 32% higher than the actual share.

#### 6.3.4 Calibrated model with new constants

The results with constants adopted from other study suggests that the accuracy of model has not been achieved. Therefore, mode-specific constant was calibrated to find the accuracy of the model fit. Probability of alternatives were predicted for the same model dataset using the same derived parameters from Table 6.2 and Table 6.3. Applying the calibrated constant into the utility functions, the difference rates between mean mode share predicted by model and actual mode share are indicated in Table 6.6, Table 6.7, and Table 6.8.

Table 6.6: Result of a test for accuracy of model fit for Homebased Other

Mode	${f Actual \ share^1}$	Predicted share	Constant	Deviation
Auto	$91.76\%^2$	91.77%	0	+0.01%
Transit	$8.24\%^{2}$	8.23%	-1.733	-0.01%
1 3 6 1 1		0)		

 $^1$  Mode share based on MiD (2008)

 $^{2}$  Modified values

Table 6.7: Result of a test for accuracy of model fit for Homebased Work

Mode	$Actual share^1$	Predicted share	Constant	Deviation
Auto	$86.03\%^{2}$	86.03%	0	0.0%
Transit	$13.97\%^{2}$	13.97%	-0.878	0.0%

<sup>1</sup> Mode share based on MiD (2008)

 $^2$  Modified values

Mode	${f Actual \ share}^1$	Predicted share	Constant	Deviation
Auto	62.63%	62.63%	0	0.0%
Transit	37.37%	37.37%	0.433	0.0%
1 1 1	· · · D		010)	

Table 6.8: Result of a test for accuracy of model fit for Homebased Airport

<sup>1</sup> Mode share to airport in Bavaria based on FMG (2016)

As shown in the Table 6.6, Table 6.7, and Table 6.8, the mode-specific constants capture the effects of the un-included attributes and measurement errors. Auto has a constant value of 0 because it is the base alternative. Transit has both negative and positive constants for commute, personal (other) and personal (airport) respectively. The calibrated constants for transit are small, which is plausible and desirable.

### 6.3.5 Estimating Route Choice with VISUM

It was decided to use equilibrium assignment to allocate the private demand on the network. This method was chosen mainly because of its simple approach. Demand matrices with auto were separated into 3 iteration layers with a fixed proportion of travel demand (i.e., 50%, 30%, and 20%). 50 iterations with 20 balancing iteration stages were exercised for each layer to allocate the trips into the road network. Besides, timetable based approach with shortest path search was attempted for travel demand with transit. This method assumed that the passengers are aware of the transit timetable and seek the shortest path to reach their destination.

#### 6.3.6 Inputs to VISUM

Prior to trip assignment procedure, average vehicle occupancy was applied to produce the person car trips. Average vehicle occupancy is determined from the hypothetical travel surveys found in MiD and shown in Table 6.9.

Average Vehicle Occupancy	Trip Purpose	Unit	
1.13	Work	person/vehicle	
2.76	Other	person/vehicle	
Source: Mobilität in Deutschland (2008)			

Table 6.9: Average vehicle occupancies according to trip purpose

In trip assignment stage, trips were modeled in the form outbound-return trip, not as one-way trip. Demand matrices of HW and HO with certain modes will be doubled in the form of outbound-return trip. It is worth stressing that the return trip of HA was not added in the form of return trip because not every trip to airport would return back in the same day. After the assignment task, the updated impedance are then fed back to mode choice in a loop iteration process. Figure 6.8 illustrates the simplified flowchart of assignment task.



Figure 6.8: Simplified flowchart of route assignment model in VISUM

#### 6.3.7 Simulation result

The assignment analysis yielded a plausible result with respects to private (PrT), but not public transport (PuT). Figure 6.9 shows the volume of PrT on network. In contrast, public transport assignment appears to misjudge the situation. An example of it is the unexpected high demand is distributed into the the link operated by regional bus even though shortest path algorithm indicated S-Bahn S1 is the shortest. The incidence error can be found in Figure 6.10. Figure 6.11 shows both private and public transport on the network. For convenient visualization, a maximum threshold of volume [Pers] has been set to both volume of PrT and PuT respectively. A threshold value and scale dimension are mentioned with all figures.



Figure 6.9: An example of PrT assignment result in VISUM - Threshold of 5,000 Pers trip with scale dimension 2



Figure 6.10: An example of PuT assignment result in VISUM - Threshold of 5,000 Pers trip with scale dimension 2



Figure 6.11: Assignment results in base scenario - Threshold of 500 Pers Trip with scale dimension 1.5 Source: Own illustration with VISUM

#### 6.3.8 Attempted solutions

In attempt to solve the modeling issues, different variations of model were applied. One of the applied solution was to re-design timetable of all line routes to airport without coordination. The simulation results did improve the public transport assignment but still high distribution of demand on links with regional buses. Even the new connection through Erding is shorter in both travel time and distance compared to bus. Other solution applied to improve the model was to re-run the simulation 20 times so that the model could find the shortest path. The results after multiple simulation appears to show slight improvement. Other than above two solutions, using shortest path approach in public transport assignment was attempted. However, this simulation took roughly 10 hours for 2,796 OD pair. This approach may be deemed as inconvenient due to its extremely time consuming. The result also did not improve very much.

As described in Section 4.2, the input data is not complete. Many system assumptions were applied for convenient use. Data was only verified through computational software (i.e., Feed Validator), and the verification was not sufficient enough to ensure the accuracy of the data. It is possible the miscalculation of public transport assignment can be fixed by a finer dataset. Also, expert assessment can allow a better evaluation of public transport assignment. As the assignment results cannot be used, only mode choice model was applied for further discussion.

# Chapter 7

# Scenario Analysis

The objective of this thesis is to assess the impact of the Erding Ring Closure on passenger flows in Munich Region. For this assessment, a mode choice and route choice model were built to realize the the project's impact on travel demand. The model built was then applied to the scenario with Erding Ring Closure infrastructure. Three scenarios in total will be discussed in this chapter. The results of the analysis of each scenario are described in this chapter.

In Chapter 1, a brief account of the proposed Erding Ring Closure is presented. As stated, the project is planned to close the S-Bahn loop for airport access. Also, it is intended to improve the connection of trans-regional service from Landshut to Mühldorf. Erding Ring Closure project comprises of Erding Ring Closure, Neufahrn Curve, and Walpertskirchen Link. However, Walpertskirchen Link is not considered in this study since it is still in design process, and no concrete information of it is available.

#### 7.1 Public transport network

The proposed rail infrastructure for scenario is incorporated into the public transport network by an allocation of new nodes and links with its respective characteristics. The development scenarios have the following characteristics: a mainline rail type, operational speed of 80 kilometers per hour, 2 tracks, and a stop point. The proposed private transport network is described in Figure 7.1. For the impact study, the a couple of system assumptions have been applied. Table 7.1 summarizes the new transit service and modified service for all scenarios. The following services run in both directions though the table indicates only one way route.

Line ID	Type	Line Routes	N. vehicle	Headway	Normal
			journeys	(Peak	Headway
				hour)	
S2	Extension	Airport-	55	$20 \min$	$20 \min$
		Peterhausen			
SEF	New	Fresing- Erding	45	$15 \min$	$30 \min$
$\mathrm{FEX}^1$	New	Landshut- Dorfen	20	$60 \min$	$60 \min$
ÜFEX	New	Landshut- Dorfen	20	$60 \min$	$60 \min$

 Table 7.1: Summary of new transit service for all scenario

<sup>1</sup> FEX operates with more intermediate stops



Figure 7.1: Erding Ring Closure Scenario Source: Own illustration with VISUM

## 7.2 Prediction

The proposed rail connection and transit services described in Section 7.1 was used to estimate the impact of Erding Ring Closure. This section discusses all future scenarios and the prediction results. To explore the impact of Erding Ring Closure, scenario with Ering Ring Closure line added is analyzed. Other two scenarios are created in addition to scenario Erding Ring Closure in order to test model sensitivity. The specifications for these two scenarios are doubled auto operating cost (AOC) and doubled parking cost (PC). It is worth noting that the scenario doubled AOC and scenario doubled PC are modeled based on the situation after Erding Ring Closure and Neufarhn Curve are put into operation. As discussed in chapter 6, all observed attributes were related to transport facilities. The model described in previous chapter is applied to predict the impact. Furthermore, as described in Section 6.3.7 and Section 6.3.8, public transport assignment results are unrealistic to the point that cannot be applied to analyze the impact of Erding Ring Closure. Thus, it was decided not to make a comparison study of both private and public transport assignment.

## 7.3 Scenario Erding Ring Closure

Key aspects being discussed in this section are categorized into three cases - air passenger mode share (HA), airport mode share (HA-HO-HW), and overall mode share in study area. To compare the results, four zones adjacent with airport were assumed to be airport zone (previously discussed in Section 5.4). Figure 7.2, Figure 7.3, and Figure 7.4 present the results predicted by model compared to scenario Erding Ring Closure according to its designated case. Keep it in mind, HA represents only air passenger trips, not the commuting trip to airport. The commuting trip to airport is covered by trip purpose - HW.

#### Air passenger mode share (HA)





The effect of Erding Ring Closure on travel mode share has been minimal, but somewhat encouraging. With the improvement of the airport access through Erding, transit is able to attract trips from the dominant auto mode. The analysis results presented in Figure 7.2 shows the mean difference of pre- and post- Erding Ring Closure. The predicted value appears to be minutest for both modes. For instance, there is a 0.07% decrease for auto in scenario Erding Ring Closure. This very small change can be referenced to the travel time. It is true that airport will have three rapid transit connections; however, the new service does not have any impacts on the daily mobility. This is due to the longer travel time with the new service. Despite the existence of new service, travel to airport with S8 still remains superior. Travel time on new service predicted by the model is 54 min from Hauptbahnhof to airport S-Bahn station. Moreover, travel time will be 34 min to reach airport if trip starts at Ostbahnhof. In contrast, travel time with S8 only takes 30 min and 41 min from Ostbahnhof and Hauptbahnhof respectively.

#### Mode share to airport (HA-HO-HW)

Regardless of trip purposes, Figure 7.3 focuses on the trip with airport as destination. The predicted shares are as follows.



Figure 7.3: Mode share to airport - Comparison between base scenario and Erding Ring Closure Scenario

Similar to air passenger mode share, the new service does not show much impact on the mode share.

#### Overall mode share in study area

Figure 7.2 indicates mean mode share in study area predicted by model compared to base scenario. The results of predicted shares are as follows.



Figure 7.4: Mode share in study area - Comparison between base scenario and Erding Ring Closure Scenario

Aside from the other two cases, the new rail corridor increase transit share by only a slight amount from 19.85% to 19.87% in the case of study area. The slightly better transit share may be due to better transit service, but not enough to draw a larger amount of share from auto.

## 7.4 Scenario double AOC

The reason to include this scenario in this thesis is mainly to test the model sensitivity. Other reason for it is the travel cost has been found to have a large influence on mode choice (Ortúzar 2000; Vuchic 2005; Ahern & Tapley 2008; Twaddle, 2011). These results suggest that increasing in auto operating cost can significantly affect the decision of individual on mode choice.

#### Air passenger mode share (HA)





This scenario proved to be effective if we want to push car users from their personal automobile. This result reflect the push-pull measure that aims to encourage traveler to use a more sustainable transport besides auto. According to Figure 7.5, auto saw a large reduction from 62.63% to 60.44%. This results also highlight the importance of cost variable in mode share analysis.

#### Mode share to airport (HA-HO-HW)

Regardless of trip purposes, Figure 7.6 focuses on trip with airport as destination. As expected, there is a significant shift from auto to transit when high auto operating cost is considered. The predicted shares are as follows.



Figure 7.6: Mode share to airport - Comparison between base scenario and Doubled AOC

#### Overall mode share in study area

Figure 7.2 indicates mean mode share predicted by model compared to base scenario. The results of predicted shares are as follows.



Figure 7.7: Mode share in study area - Comparison between base scenario and Doubled AOC

In contrast, the overall mode share does change but with a slight amount. However, the result is statistically significant.

### 7.5 Scenario double Parking Cost

The decision to include this attribute is because this variable is so responsive to policy change. The scenario with doubled parking cost only applied to commuting and other trip, not the airport trip. It may be not rational to double the airport parking fee as its parking places are designed to accommodate thousands of cars. General parking in city center is different as it is always high in demanded. Thus, increasing parking cost in general parking cost is more reasonable than airport parking.

#### Mode share to airport (HA-HO-HW)

Regardless of trip purposes, Figure 7.2 focuses on trip with airport as destination. The predicted shares are as follows.



#### MODE SHARE TO AIRPORT - DOUBLED PC

Figure 7.8: Mode share to airport - Comparison between base scenario and Doubled PC

As expected, the results can be in turn explained by the burdensome of high operating cost. With auto becomes more expensive to use, traveler may consider using transit. This is evident by the results predicted. There is an obvious rise in transit when parking cost is doubled.

#### Overall mode share in study area

Figure 7.2 indicates mean mode share predicted by model compared to base scenario. The results of predicted shares are as follows.



Figure 7.9: Mode share in study area - Comparison between base scenario and Doubled PC

The analysis results shows that increasing in parking cost does not have impact as large as increasing in auto operating cost. Although the shift in mode share appears to be small in value, at least, this policy change can push a few travelers to use transit.

#### 7.6 Summary

The predicted share for base case scenario are indeed different from the mode share provided by MiD. The share in this chapter are predicted as a mean value without taking the trip purpose into account. Further, it is also worth noting that only attributes relative transport system (time and cost components) are considered. As one would expect, auto shares are predicted to have the larger share compared to transit. Auto share still outperforms the transit though transit

becomes more attractive. This is perhaps because of the high level of comfort, convenience, and flexibility of auto. Table 7.2 summarizes only the predicted share in study area regardless of trip purpose. The results are based on Figure 7.4, Figure 7.7, and Figure 7.9.

Table 7.2: Mode share for trips in Munich region by scenario

Mode	$\mathrm{Base}/\%$	$\mathrm{ERC}/\%$	Doubled AOC/%	Doubled PC
Auto	80.15%	80.13%%	79.21%	80.06%
Transit	19.85%	19.87%%	20.79%	19.94%

# Chapter 8

# Conclusion

This thesis aims to assess the impact of Erding Ring Closure on passenger flows in Munich region. The study focused only on two alternatives - auto and transit as these two are deemed the most suitable for regional scale study. To help select models, literature and previous studies relevant to this topic was studied. Consequently, binary logit choice model based on discrete choice theory was built and applied to scenarios to predict the changes in modal split. Apart from mode choice exercise, an attempt to find out whether an addition of new infrastructure and service can affect the present pattern of route choice was made. The study analyzed assignment of mode-specific trips on network by using equilibrium assignment for private car and timetable-based approach for public transport. Furthermore, this thesis used the computational software to aid the analysis process because of its ability to model the travel demand as realistic as possible and time-saving. PTV VISUM, a macroscopic traffic simulation software, was employed to model the mode choice and simulate route assignment.

The attributes used to specify the mode choice model was limited to only the objective variables that fall under the category of transport facility. While estimating the model parameters is not a simple task in the case of limited data availability, the study adopted derived parameters from other study to formulate mode choice. The coefficients presented in the analysis have proved to be viable for the study. As expected, the analysis with derived parameters yielded a reasonable result for airport trip, which is by definition a long-distance travel trip. On the other hand, the coefficients, however, seemed not to be feasible with short-distance travel trip - commuting and other trip. Despite these results, mode-specific constants still needed to be calibrated to reflect the actual mode share of all trip purposes. Model accuracy test was undertaken to match the actual mode share reported by MiD and of airport. The calibration indicated desirable small constants for transit in all trip purposes.

To find out if Erding Ring Closure has an impact on travel mode, three scenarios were performed. The results of scenarios with Erding Ring Closure presented the minimal impact of the new service on modal split. The impact has not been wide as expected; however, the impact was encouraging and sobering. The addition of new connection was able to draw passenger from auto. Nevertheless in overall, auto still dominated the mode share regardless of trip purposes. Another two scenarios attempted to analyze the impact of high auto operating cost and high parking cost. These two scenarios were modeled with the existence of Erding Ring Closure. Scenario with high auto operating cost highlighted significant decrease in auto use compared to base scenario. More importantly, double operating cost has led to remarkable reduction in auto share to airport. This signifies the importance of operating cost variable compared to travel time. The last scenario, whose parking fee is assumed to be doubled, did not add much competition to auto. The resulting model seems to explain that increasing in parking cost has less impact than increasing auto operating cost. Another topic addressed in this thesis is the analysis of route assignment. For the analysis of route choice, only base scenario was attempted. Through user equilibrium, results revealed a fair distribution of passenger volume on the link. On the transit side, the assignment of public transport made use of timetable-based approach to assign the trips on the routes. Unfortunately, the analysis did not yielded neither accurate nor plausible results of public transport assignment. Assignment result exhibited unrealistic situation for base scenario. For this reason, thorough scenario analysis was not able to perform for public transport. As one would expect, Erding Ring Closure aims to provide a better alternative to transit, not auto. A discussion on private assignment would be not much of interest.

## 8.1 Concluding remark

This thesis has partially achieved the objective that was set. The results from the analysis are statistically significant though they are minimal. Each scenario encompasses three different aims, but all endeavor to forecast the potential impact as a result of new infrastructure, new service, and policy changes. Scenario with Erding Ring Closure highlights the importance of time variable. Though the connection to airport has been improved, auto outperforms the transit. The other two scenario attempt to investigate any impact in response to policy change. The results are evident by a decrease in auto share regardless of trip purposes. Additionally, the results presented in the last two scenarios appear to contradict some previous studies' claim (Moeckel et al., 2015), which stated that parking cost was perceived as more onerous than auto operating cost. Last but importantly, the overall results evidently indicate that airport trip is susceptible to the influence of Erding Ring Closure.

Besides, the study emphasizes the importance of input dataset. The results derived from the analysis are dependent on the level of accuracy of the dataset. The miscalculation in public transport assignment stage could be explained by the poor timetable information. Timetable-based approach relies on a precise departure and arrival time of public transport. With reliable timetable, this approach takes into account of a coordination in order to yield a very precise result. However, the loose timetable information has misled the assignment, and thus makes it misjudge the situations. Moreover, the current dataset is not complete, and this requires a significant improvement of dataset. This error in dataset can be amended through expert assessment and more experiment tests.

The use of computational software for in-dept analysis has proved to be effective and efficient. VISUM makes many contribution to the analysis process. VISUM's ability to model realistic travel demand is evident and proved. The results are also feasible and logical according to input data used.

Overall, I argue that this work is novel and this study requires further researches to help improve the validity and accuracy of the results.

## 8.2 Limitations of the study

There are several limitations that should be addressed to make ways for future work to improve the reliability and validity of predicted results. Some of the limitations are described as follow.

Modeling a large scale truthful travel demand is not a simple task. The model requires a significant amount of input data, which must be as accurate as possible. The input data used to build model in VISUM are entirely made from assumptions. GTFS was created from scratch, applying many assumptions that do not reflect the reality, mainly for bus, regional bus, and

regional train. This has resulted in inaccurate timetable and poor route information for public transport. Should GTFS be utilized for future research, a correcting work in source dataset is required.

Another limitation of the model is the adjustment of mode-specific constants with MiD data. Mode-specific constants for home-based work and home-based other trip were calibrated to match the MiD mode choice pattern which is the actual share from all urban areas in Germany. Performing the calibration with information exclusive to the study area would have provide a better and more accurate result.

Constrained by the lack of data available, the study did not cover the non-home-based (NHB) trip. NHB trip normally accounts for roughly 25% to 30% of daily mobility by individuals in urban areas (Schultz & Allen, 1996). Extending the study with NHB trip could not only contribute to better urban travel model but could also increase the sensitivity and accuracy of the forecast.

Setting in a future time frame, travel demand of today may not be deemed convenient to describe the future travel demand with Erding Ring Closure. One favored solution of it can be achieved through the application of expansion factor or the use of demand growth rate that is derived from empirical analysis.

Last but importantly, model was not calibrated and validated before it was applied to scenarios. Thus, it is difficult to determine the validity of the results.

### 8.3 Suggestion for future researches

The models described in this study has been proved to be suitable with mode share analysis, however, not for route assignment task. The model failed to yield thorough and plausible results for public transport assignment. Hence, this study could be a good starting point for future work to better understand the causes of the incidences. Future researches are recommended to help improve this study by addressing the limitations mentioned above.

Moreover, in this study, the variable included in mode choice model only address the attributes of transport facility. It would be interesting, if future works can also extend their focuses to other explanatory variables such as individual related attributes and trip characteristic. This can perhaps predict more precise results.

Furthermore, the study used equilibrium assignment to assign person car trips on the network. Performing the assignment with a more sophisticated approach like Stochastic Equilibrium could also explain a realistic situation of passenger flows.

A detailed discussion on travel time and relative accessibility to Messe München is recommended. The research on these two aspects may help realize the impact of Erding Ring Closure.

Additionally, the study geographically distributed the passenger survey to the level of resolution of TAZ. Passenger survey only reflects the distribution at the time of the survey and may not reflect the future demand. The use of passenger survey data as distribution model may not provide a suitable level of resolution of air passenger trip origins. It would be interesting if airport trip generation is modeled instead of exogenously incorporating passenger survey data into the demand model. Previous studies on airport trip generation have been conducted by California High-Speed Rail system (CSI et al., 2006), Gosling et al. (2003), and Gosling (2011).

# Bibliography

- [1] Ahern, A. A. & Tapley, N. (2008). The use of stated preference techniques to model modal choices on interurban trips in Ireland. Transportation Research Part A. Vol 42. pp. 15-27.
- [2] Alliance Transportation Group (2015). AHTD National County-Level Long Distance Travel Model: Model Development and Validation Report. 15th Transportation Research Board National Transportation Planning Applications Conference
- [3] Allgemeiner Deutscher Automobil-Club e.V. [ADAC] (2017). ADAC Autokosten Herbst/Winter 2017/18 [Auto Cost for Autumn/Winter 2017/18] Retrieved from: https://www.adac.de/\_mmm/pdf/autokostenuebersicht\_47085.pdf
- [4] Axhausen, K. W. & Polak, J. W. (1991). Choice of parking: Stated preference approach. Transportation. Vol 18. pp. 59-81.
- [5] Basar, G. & Bhat, C. (2004). A parameterized consideration set model for airport choice: an application to the San Francisco
- [6] Bayerisches Staatminesterium des Innern, für Bau und Verkehr (STMI) (2010). Erdinger Ringschluss [Erding Ring Closure].
   Retrieved from: http://www.stmwivt.bayern.de/fileadmin/Web-Dateien/ Dokumente/verkehr/ringschluss/Erdinger\_Ringschluss\_Praesentation\_100727.pdf
- Bazghandi A. (2012). Techniques, Advantages and Problems of Agent Based Modeling for Traffic Simulation, *IJCSI International Journal of Computer Science Issues*. Vol. 9. Issue 1. No 3. January 2012
- [8] Ben-Akiva, M., & Lerman, S. R. (1985). Discrete choice analysis: Theory and applications to travel demand. Cambridge: MIT Press.
- [9] Blackstone, E. A., Buck, A. J. & Hakim, S. (2006). Determinants of airport choice in a multi-airport region, Atlanta Economic Journal. 34. 313–326.
- Brodowsky, K. (2013). Bedeutung der öffentlichen Verkehrsmittel Retrieved November 14, 2017 from: https://karl.brodowsky.com/2013/03/21/bedeutung-der -offentlichen-verkehrsmittel/
- [11] Bundesagentur für Arbeit (2016). Stastistik
- [12] Cambridge Systematics, Inc. (CSI). Bay Area/California high-speed rail ridership and revenue forecasting study: inter regional model system development, Prepared for the Metropolitan Transportation Commission and the Calif. High-Speed Rail Authority. Oakland. California.
- [13] Chow, H. F. A. (2007). Trip Assignment a literature review. California PATH UC Berkeley

- [14] Djoko, E. (2014). Evaluation of the feasibility of a new North-South Metro line in Stockholm from an infrastructure and capacity perspective. Master Thesis. KTH ROYAL INSTITUTE OF TECHNOLOGY
- [15] Frei, A. (2008). Survey issues in long-distance travel. Conference paper STRC 2008
- [16] Forinash, C. V., & Koppelman, F. (1993). Application and Interpretation of Nested Logit Models of Intercity Mode Choice. TRANSPORTATION RESEARCH RECORD 1413
- [17] Gosling, G. D. (2003). SCAG Regional Airport Demand Model: literature review, Prepared for the Southern California Association of Governments. Los Angeles. California.
- [18] Gosling, G. D. (2011). Ground Access Trip Generation Models for Airport Planning
- [19] Gupta, S. (2010). Stochastic User Equilibrium. CE682. Infrastructure and Transportation Planning
- [20] Twaddle, H. A. (2011). Stated preference survey design and pre-test for valuing influencing factors for bicycle use. Master's thesis. Technical University of Munich
- [21] Hess, S. & Polak, J. W. (2005). Mixed logit modelling of airport choice in multi-airport regions. Journal Air Transport Management. Vol 11. 59–68.
- [22] Hess, S. & Polak, J. W. (2006a). Airport, airline and access mode choice in the San Francisco Bay area. Pap. Reg. Sci. Vol 85. 543–567.
- [23] Hess, S. & Polak, J. W. (2006b). Exploring the potential for cross-nesting structures in airport-choice analysis: a case-study of the Greater London area, Transp. Res. Part E Logist. Transp. Rev. Vol 42. 63–81.
- [24] Hoogendoorn, S. P., & Bovy, P. H. L. (2004). State-of-the-art of Vehicular Traffic Flow Modelling. Special Issue on Road Traffic Modeling and Control of the Journal of Systems and Control Engineering. Delft University of Technology
- [25] Institution of Highways and Transportation (1997). Transport in the Urban Environment.
- [26] Ji, J. Y. (2017). Long-Distance Mode Choice Modeling of Ontario Province, Master thesis, Technical University of Munich
- [27] Kirchhoff, P. (2002). Städtische Verkehrsplanung. Teubner Verlag. Stuttgart u.a.
- [28] Knoflacher, H. (2015). Kurzdokumentation des generationsübergreifenden Verkehrskonzepts "Zukunft am Lech"
- [29] Koppelman, F. S., & Bhat, C. (2006). A Self Instructing Course in Mode Choice Modeling: Multinomial and Nested Logit Models. Retrieved from U.S. Department of Transportation, Federal Transit Administration website: Retrieved from: http://www.caee.utexas.edu/prof/bhat/COURSES/LM\_Draft\_060131Final-060630.pdf
- [30] Li, S. (2015). Influence of a New Subway Line's Opening on Passenger Flow Characteristics of an Urban Rail Transit Network. 15th COTA International Conference of Transportation Professionals, 1757.
- [31] Lohse, D., & Schneider, R. (1997). Vergleichende Untersuchungen der aggregierten und disaggregierten Verkehrsplanungsmodelle. Schriftenreihe des Instituts für Verkehrsplanung und Straßenverkehr. Heft 3/1997. Dresden.
- [32] Mathew, T. V., & Rao, K. V. K. (2007). Chapter 10: Traffic Assignment Retrieved from: http://nptel.ac.in/courses/105101087/downloads/Lec-10.pdf
- [33] McFadden, D. (1974). Conditional logit analysis of qualitative choice behavior. In P. Zarembka (ed.), Frontiers in econometrics. New York: Academic Press, 105–142.
- [34] McNally, M. G. (2007). The Four Step Model. University of California. Irvine.
- [35] Metropolregion München (2015). Retrieved November 8, 2017 from: https://www.metropolregion-muenchen.eu/metropolregi on-muenchen/daten-und-fakten/
- [36] Moeckel, R. and Donnelly, R. (2015). Gradual rasterization: redefining spatial resolution in transport modelling. Environment and Planning B: Planning and Design. 42(5). pp. 888–903. doi: 10.1068/b130199p.
- [37] Moeckel, R., Fussell, R., & Donnelly, R. (2015). Mode choice modeling for long-distance travel. Transportation Letters. 7(1). 35-46. doi:10.1179/1942787514y.0000000031
- [38] Müller, M. (2010). Design and implementation of a V2X-based dynamic routing assistance for the microscopic traffic simulation VISSIM. Master thesis. Technical University of Munich
- [39] Münchner Verkehrs- und Tarifverbund GmbH (2010a). Ergebnisbericht MiD 2008. München und Münchner Umland
- [40] Münchner Verkehrs- und Tarifverbund GmbH (2010b). Möbilität in Landkreis Dachau
- [41] Münchner Verkehrs- und Tarifverbund GmbH (2010c). Möbilität in Landkreis Ebersberg
- [42] Münchner Verkehrs- und Tarifverbund GmbH (2010d). Möbilität in Landkreis Erding
- [43] Münchner Verkehrs- und Tarifverbund GmbH (2010e). Möbilität in Landkreis Freising
- [44] Münchner Verkehrs- und Tarifverbund GmbH (2010f). Möbilität in Landkreis Fürstenfeldbruck
- [45] Münchner Verkehrs- und Tarifverbund GmbH (2010g). Möbilität in Landkreis München
- [46] Münchner Verkehrs- und Tarifverbund GmbH (2010h). Möbilität in Landkreis Starnberg
- [47] National Cooperative Highway Research Program (1998). Report 365, National Academy Press. Washington D.C.
   Retrieved from: http://onlinepubs.trb.org/onlinepubs/nchrp/nchrp\_rpt\_365.pdf
- [48] NIATT Lab Manual (2008). Cross-Classification. Retrieved from: http://www.webpages.uidaho.edu/niatt\_labmanual/Chapters/traveldemand forecasting/theoryandconcepts/CrossClassification.htm
- [49] Nixon H., Boarnet M., Houston D., Spears S., and Lee J. (2015). Changes in Transit Use and Service and Associated Changes in Driving Near a New Light Rail Transit Line. Mineta Transport Institute
- [50] Olsson A. L. (2003). Factors that influence Choice of travel mode in major urban areas. Kugel Tekniska Högskolan.
- [51] Ortúzar, J. d. D. (2000). Modelling route and multimodal choices with revealed and stated preference data. in JDD Ortúzar (ed.). Stated Preference Modelling Techniques. 4th edn, PTRC Education and Research Services Ltd.. Londres.

- [52] Ortúzar, J. d. D., & Willumsen, L. G. (2011). *Modelling Transport (Fourth Edition)*. Chichester, West Sussex, United Kingdom: John Wiley & Sons, Ltd.
- [53] Papola, A., & Marzano, V. (2006). How can we trust in the OD matrix correction procedure using traffic count?. Dipartimento di Ingegneriadei Trasporti. Università di Napoli "Federico II"
- [54] Pels, E., Nijkamp, P. & Rietveld, P. (2001), Airport and airline choice in a multi-airport region: an empirical analysis for the San Francisco bay area, Regional Studies 35(1), 1–9.
- [55] Pels, E., Nijkamp, P. & Rietveld, P. (2003), Access to and competition between airports: a case study for the San Francisco Bay area, Transportation Research Part A: Policy and Practice 37(1), 71–83.
- [56] Piatkowski, B., & Maciejewski, M. (2013). Comparison of traffic assignment in VISUM and transport simulation in MATSIM. *Transport Problem*. Vol 8. Issue 2
- [57] Planungsverband Äußerer Wirtschaftsraum München (2015). Retrieved December 1, 2017 from: http://www.pv-muenchen.de/index.php?id=0,352
- [58] PTV AG, (2017). PTV VISUM Manual
- [59] Reimann, M. (2007). Simulationsmodelle im Verkehr, Seminararbeit, Universität Karlsruhe
- [60] Schultz, G. W. & Allen, W. G. (1996). Improved modeling of non-home-based trip. Transportation Research Record 1556, 22-26
- [61] Siemen AG (2012). Light Rail Impact Study in Turku Retrieved from: https://www.turku.fi/sites/default/files/atoms/files/light\_rail\_impact\_study.pdf
- [62] Skarphedinsson, A. (2009). Evaluating a simplified process for developing a four-step transport planning model in VISUM, Master thesis
- [63] Schiffer (2012)
- [64] Schnabel, W., Lohse, D. (2011). Grundlagen der Straßenverkehrstechnik und der Verkehrsplanung Band 2: Verkehrsplanung. Beuth Verlag. Berlin.
- [65] Statistischer Ärmter des bundes und der L\u00e1nder (2016). Retrieved january 10, 2018 from: https://www.statistik-bw.de/Statistik-Portal/de\_jb16\_jahrtab36.asp
- [66] University of Kentucky (n.d.). College of Engineering: rsouley(CE451 LECTURES). Retrieved from: https://web.engr.uky.edu/~rsouley
- [67] US Bureau of Public Roads (1964). Retrieved from: http://onlinebooks.library.upenn.edu/webbin/book/lookupname?key=Unit ed%20States%2E%20Bureau%20of%20Public%20Roads
- [68] Victor, D. J., & Ponnuswamy, S. (2012). Urban Transportation: Planning, Operation and Management.
- [69] Weiner, E. (1997). Urban transportation planning in the United States: an historical overview 5th Edition. Washington D.C.: U.S. Dept. of Transportation.
- [70] Weiner, E. (1999). Urban transportation planning in the United States: an historical overview 7th Edition. Washington, D.C.: U.S. Dept. of Transportation.

- [71] Wermuth, M., Sommer, C., & Sven Wulff, S. (2006). Erhebung der individuellen Routenwahl zur Weiterentwicklung von Umlegungsmodellen. Institut f
  ür Verkehr und Stadtbauwesen. Heft 136
- [72] Zenina, N., & Borisov, A. (2013). Regression Analysis for Transport Trip Generation Evaluation. Information Technology and Management Science
- [73] Zheng, L. J. (2008). Study of Passenger Flow Distribution Prediction based on URT Network Operation, Master thesis. Shanghai: Tongji University.
- [74] VCD (n.d.). Verkehrsmittel im Vergleich [Transportation in Comparison] Retrieved December 1, 2017 from: https://www.vcd.org/themen/klimafreundliche-mobilitaet /verkehrsmittel-im-vergleich/
- [75] Vuchic, V. R. (2005). Urban Transit Operations, Planning and Economics. Wiley. John & Sons. Incorporated.

### Appendix A

Mode choice modelling for long-distance travel conducted by Moeckel, Fussel, and Donnelly (2015)

Derived parameters used for the mode choice model

Table 1 Coefficients and variables for utilities of drive-alone and shared-ride mode
--

Coefficient	Description	Value
u <sub>i,jm,p</sub>	Utility of mode <i>m</i> (drive-alone, shared-ride 2, shared-ride 3, or shared-ride 4+) from zone <i>i</i> to zone <i>i</i> for purpose p	
ivtc	In-vehicle time coefficient: parameter evaluating the travel time spend in the vehicle	(b) –0.026, (p) –0.018, and (c) –0.025*
tt <sub>i,i</sub>	Travel time from zone i to zone j by auto	
ovtc	Out-of-vehicle time coefficient: parameter evaluating the travel time spend out of the vehicle	(b) -0.052, (p) -0.036, and (c) -0.050*
autoEgr	Auto-egress time: time spent to walk from the parking location to the final destination	5 min
prkc <sub>p</sub>	Parking costs coefficient by purpose p Parking costs in zone i	(b) -0.006, (p) -0.012, and (c) -0.010*
occm	Number of persons traveling by drive-alone, shared-ride 2 shared-ride 3 and shared-ride 4+	1, 2, 3, 4·1 persons
aocc <sub>p</sub>	Auto operating costs coefficient by purpose Distance from zone <i>i</i> to zone <i>i</i> in miles	(b) -0.004, (p) -0.008, and (c) -0.007*
aoc	Auto operating costs	0.0874 cents mile <sup>-1</sup>

\*Trip purposes: (b) business, (p) personal, and (c) commute; in line with nested mode choice theory, parameters are scaled by the nesting coefficient.

Coefficient	Description	Value
u <sub>i,j,m,p</sub>	Utility of mode m (bus, rail, or air) from	
	zone i to zone j for purpose p	
ivtc	In-vehicle time coefficient: parameter evaluating	(b) -0.026, (p) -0.018, and (c) -0.025
	the travel time spend in a vehicle	
tt <sub>i j, m</sub>	Travel time from zone <i>i</i> to zone <i>j</i> on mode <i>m</i>	
ovtc	Out-of-vehicle time coefficient: parameter	(b) −0.052, (p) −0.036, and (c) −0.050 <sup>*</sup>
	evaluating the travel time spend out of the vehicle	
trnAccm	Time to access the transit, which includes the walk	Bus: 15; rail: 30; and air: 60
	from the vehicle to the transit platform or gate, time	
	for check-in and time for security checks	
ntrc	Coefficient on number of transit transfers	-0·01*
trnsf <sub>iStat</sub> jStat,m	Number of transfers to travel from iStat to jStat on mode m	
trfcp	Transit fare coefficient by purpose	(b) -0.006, (p) -0.012, and (c) -0.010*
fare <sub>iStat.iStat.m</sub>	Transit fare from iStat to jStat on mode m	
tfqcp	Coefficient on frequency of service per day by mode	(b) 6, (p) 2, and (c) 3*
frquiStat, Stat.m	Frequency of service from iStat to jStat on mode m per day	
dist	Distance from zone i to zone j	
trnEgrm	Time to egress the transit, which includes the walk from	Bus: 10; rail: 15; and air: 20
	the transit platform or gate to a vehicle and time	
	for collecting baggage	

Table 2 Coefficients and variables for utilities of transit modes bus, rail, and air

\*Trip purposes: (b) business, (p) personal, and (c) commute; in line with nested mode choice theory, parameters are scaled by the nesting coefficient.

Table 4 Comparison of parame	ters across l	ong-distance	logit-based mode	e choice models					
Author	Grayson (19	81)		manifesting and	Bhat (1995) <sup>2</sup>	Baik et al. (20	(80	tiopic	Milicon
Purpose	Business	Social	Entertainment	(2001) <sup>1</sup>	Business	Business	Non-business	Miller (1982)	wiison et al. (1990)
Sample size	2720 (NTS 1	(779		4324	2769	540000 (ATS 1995)			1624
Cost	-0.0328	-0.0111	-0.0111	-0.0173	-0.0318	-0.0309	-0.0275 to		
Auto-operating costs Fare/income						0 - 0.0034	-0.0181	-0.3507	
Travel cost/distance In-vehicle time	-0.02	-0.238	-0-079	-0.0031	-0.011	-0-0269 to -0.2608	-0.1219 to	-0.1075 to	- 15-084
In-vehicle time for rail and air In-vehicle time for bus and auto In-vehicle time for auto In-vehicle time for rail									
In-venicle time for air Travel time/distance Out-of-vehicel time, wait	-0.0244	600.0-	-0.003	-0.011	-0.0362				- 166·285
Frequency Frequency / distance		0 0 0	0	0.0288	0.0741				0-018
Transfers Frequency for rail and bus									
Access	-0.0006	-0.0016	-0.0015					-0.01442	
Access for rail and bus Access for air									
Egress air Egress rail Egress bus								0-0621 -0-004578 -0-03612	
Constant auto	0	0	0	5.133	0				
Constant bus	-3.159	-3.016	-1.797	0	N/A				16-596
Constant rail	-2:461	-2.828	-3·288	4-981	-0.1763			0.2032	18-016
Constant air	-1.313	-3.166	-3.435	6.264	-0.4883			-0.3391	15-382
Constant large city train Constant large city air					1-9066 0-7877				
Constant Income train Constant income air					-0.0167 0.0223				
Constant manufacturing								0-8612	
employment Constant finance and								0.2715	
service employment Constant medical &								0-6755	
aovernmment employment									

#### Moeckel et al. Mode choice modeling for long-distance travel

Moeckel et al. Mode choice modeling for long-distance travel

### Appendix B

### Network modeling in VISUM

#### Task 1: Demand Model

emand models	5				
elect demand m	odel				
01 Model 1		~			
Basis Person gr	oups Activity pairs	Demand strata			
Number: 3	Code	Name	Person groups	Activity pair	
1	HA	Home Airport	POP	HA Home-Airport	
2	HO	Home Other	POP	HO Home-Other	•
3	HW	Home Work	EMP	HW Home-Work	•
Create	Delete		Productions/attra	actions	
				ОК	Cancel

### Task 2: Transport System Modes

TS	ys/Modes/DS	eg			X
т	ransport system	ns Modes D	emand segment	S	
	Number: 2	Code	Name	Mode	TSys
S.		С	Car	С	CAR
	2	PT	PT	PuT	BUS,LR,PUT
	<u>C</u> reate	<u>E</u> dit	<u>D</u> elete		
				OK	Cancel

#### Task 3: Importing Matrices

Add external matrix to the network model X					
Number	Number 1				
Code	Distribution HA				
Name	Distribution HA				
<ul> <li>Demand ma</li> <li>Skim matrix</li> <li>Demand ma</li> </ul>	trix for zones for zones trix for main zones				
◯ Skim matrix for main zones					
File:					
C:\Users\Rena	Nong\Documents\Thesis\VISUM\9 Distribution HA.m				
OK Cancel					

### Task 4: Importing OSM Data

OpenStreetMap Importer >							
OSM-Files	studyarea2.osm						
Configuration	Detailed urban network			$\sim$			
Description	Converts the road network i transport network. Suited for	including footpaths and cycleway or urban models.	rs and the full public	^ ~			
Retain tempora ry *.net	C:\Users\Rena Nong\AppDat	ta\Local\Temp\2018-03-07_1121;	21.net				
Clip networ k to boundi ng box	48.2963165       11.5513409       48.0727356						
	Import OSM-elements which are completely inside the bounding box						
	O Import OSM-elements which are completely or partially inside the bounding box						
	Reset						
I have read accept these http://www.	the OpenStreetMap data lice guidelines openstreetmap.org/copyright	ensing terms and	ОК	Cancel			

### Task 5: Importing Zone

Read shapefile	×
Name of the shapefile Number of objects in the shape file Read additively	sa_2_taz.shp 2796
Read as	Zones
Offset	0
Normalize created and modified p	olygons (self-intersecting test)
Merge points with identical co-ord	inates
Fuzzy surface alignment	
Tolerance for imported surfaces	10.00m
Reference surfaces	
	OK Cancel

Task 6: Setting Connector (Example of PrT Connector) - 5 km is set for direct distance

Generate connectors	×
Only active zones / nodes are connected	
● PrT ○ PuT	
Maximum length (direct dist.)	1.000km
Maximum number (current step)	1
Maximum total number per zone (PrT)	1
Type of generated connectors	9
ОК	Cancel

### Task 7: Importing GTFS

Open Direc	tory	Open Zip-File
Sele dd.r	ect a date nm.yyyy a	between nd dd.mm.yyyy
	ОК	Cancel
	Open Direc	Open Directory Select a date dd.mm.yyyy a OK

#### Task 8: Calculate Skim Matrix for PrT

Parar	meters: PrT sk	im matrices				Х		
Cons	idered OD pain	s						
Г	Calculate only	OD pairs with demand	> 0 all		$\checkmark$			
Path	selection							
P=	th search crite	rion	Impedan	~P	~			
			Impedant					
$\checkmark$	Use paths fro	m assignment						
W	eighting of pat	hs	Mean ove	Mean over path volume $\vee$				
Sum	up paths from	1						
	Links	✓ Turns	Origin connectors	🗹 Destinatio	on connectors			
Chim								
SKIM	15		1					
Ν	Number: 13	Calculate	Save to file	Open	Skim	^		
	1				t0			
	2				tCur			
	3				v0			
	4				vCur			
	5	×			Impedance			
	6				Trip distance			
	/				Direct distance			
	8							
	9					-		
	10				AddValue-TSvs			
	12				Toll	_		
	12				10			
Outp	out file							
Fil	le name							
Fo	ormat	Format V 🛛 🗸	rmat V $\sim$ Means of transport no. (Tour-based model) 🗹 4					
Se	eparator	rr Blank V Confirm overwriting 🗹						
					OK Cance	el		

Task 9: Calculate Skim Matrix for PuT

Basis   Search   Preselection Impedance   Choice   Skim matrices   Iterations   Vol/cap ratio dependent   • •         Perceived journ. time PJT =         be Coefficient Attribute   1.0   1.00			т:	4 - I-						$\sim$
Basis   Search   Preselection Impedance   Choice   Skim matrices   Iterations   Vol/cap ratio dependent i ◆ ◆         Perceived journ. time PJT =         be Coefficient Attribute * 1.0         1.00       In-vehicle time * 1.0         1.00       PuT-Aux ride time * 1.0         1.00       PuT-Aux ride time * 1.0         1.00       Access time         1.00       Corrigin wait time         Parameters       1.00         + 1.00       Origin wait time         Parameters       1.00         + 1.00       Transfer wait time         Parameters       1.00         + 0.00       Extended impedance         Parameters       1.00         + 0.00       Fare         1.00       PJT [min]         1.00       Intr	aram	neters: Assi	gnment procedure: Time	tab	le-based					~
be Coefficient       Attribute       BoxCox       Lambda         1.00       In-vehicle time       *       1.0       1.00         +       1.00       PuT-Aux ride time       *       1.0       1.00         +       1.00       PuT-Aux ride time       *       1.00       1.00         +       1.00       Access time       Introduction       1.00         +       1.00       Egress time       Introduction       1.00         +       1.00       Origin wait time       Parameters       Introduction         +       1.00       Origin wait time       Parameters       Introduction         +       1.00       Transfer wait time       Parameters       Introduction         +       1.00       Transfer wait time       Parameters       Introduction         +       0.00       Extended impedance       Parameters       Introduction         •       0.00       Extended impedance       Parameters       Introduction         •       0.00       Extended impedance       Introduction       Introduction         •       0.00       Fare       Introduction       Introduction         •       0.00       Fare       Introduction	Basis   Search   Preselection Impedance   Choice   Skim matrices   Iterations   Vol/cap ratio dependent i									•
1.00       In-vehicle time       *       1.0       1.00         +       1.00       PuT-Aux ride time       *       1.0       1.00         +       1.00       Access time       1.00       1.00         +       1.00       Egress time       1.00       1.00         +       1.00       Egress time       1.00         +       1.00       Usational time       Parameters       1.00         +       1.00       Origin wait time       Parameters       1.00         +       1.00       Transfer wait time       Parameters       1.00         +       1.00       Transfer wait time       Parameters       1.00         +       0.00       Extended impedance       Parameters       1.00         +       0.00       Fare       1.00       1.00         +       0.00       Fare       1.00       1.00         +       1.00       DeltaT(late) [min]       1.00	he	Coefficient	Attribute				BoxCox	Lambda		
+       1.00       PuT-Aux ride time       *       1.0       1.00         +       1.00       Access time       1.00       1.00         +       1.00       Egress time       1.00       1.00         +       1.00       Walk time       1.00       1.00         +       1.00       Origin wait time       Parameters       1.00         +       1.00       Origin wait time       Parameters       1.00         +       1.00       Transfer wait time       Parameters       1.00         +       1.00       Transfer operator cha       Parameters       1.00         +       0.00       Extended impedance       Parameters       1.00         +       0.00       Extended impedance       Parameters       1.00         -       0.00       Extended impedance       Parameters       1.00         -       1.00       Extended impedance       Parameters       1.00         -       1.00       PJT [min]       1.00       1.00         +       0.00       Fare       1.00       1.00         +       1.00       DeltaT(late) [min]       1.00       1.00         +       1.00       DeltaT(late		1.00	In-vehicle time	*	1.0			1.00		
+       1.00       Access time       □       1.00         +       1.00       Egress time       □       1.00         +       1.00       Walk time       □       1.00         +       1.00       Origin wait time       Parameters       □       1.00         +       1.00       Origin wait time       Parameters       □       1.00         +       1.00       Transfer wait time       Parameters       □       1.00         +       1.00       Transfer wait time       Parameters       □       1.00         +       0.00       Transfers *       Formula       □       1.00         +       0.00       Extended impedance       Parameters       □       1.00         +       0.00       Extended impedance       Parameters       □       1.00         -       -       Oscider connections with DeltaT > 0, if connection with DeltaT = 0 exists       Impedance =         -       -       1.00       PJT [min]       □       1.00         +       0.00       Fare       □       1.00       1.00         +       1.00       DeltaT(late) [min]       □       1.00         +       1.00 <t< td=""><td>+</td><td>1.00</td><td>PuT-Aux ride time</td><td>*</td><td>1.0</td><td></td><td></td><td>1.00</td><td></td><td></td></t<>	+	1.00	PuT-Aux ride time	*	1.0			1.00		
+       1.00       Egress time       □       1.00         +       1.00       Walk time       □       1.00         +       1.00       Origin wait time       Parameters       □       1.00         +       1.00       Transfer wait time       Parameters       □       1.00         +       1.00       Transfer wait time       Parameters       □       1.00         +       0.00       Transfer of operator cha       Parameters       □       1.00         +       0.00       Extended impedance       Parameters       □       1.00         DeltaT =       Time difference between desired and actual departure or arrival time       □       □       □         Consider connections with DeltaT > 0, if connection with DeltaT = 0 exists       Impedance =       □       □       □         beCoefficient       Attribute       BoxCox       Lambda       □       □       □       □       □	+	1.00	Access time					1.00		
+       1.00       Walk time       -       1.00         +       1.00       Origin wait time       Parameters       -       1.00         +       1.00       Transfer wait time       Parameters       -       1.00         +       1.00       Transfer wait time       Parameters       -       1.00         +       1.00       Transfer wait time       Parameters       -       1.00         +       0.00       Transfer of operator cha       Parameters       -       1.00         +       0.00       Extended impedance       Parameters       -       1.00         +       0.00       Extended impedance       Parameters       -       1.00         -       -       -       -       -       -       -         -       Consider connections with DeltaT > 0, if connection with DeltaT = 0 exists       Impedance =       -	+	1.00	Egress time					1.00		
+       1.00       Origin wait time       Parameters       1.00         +       1.00       Transfer wait time       Parameters       1.00         +       2min       Number of transfers       *       Formula       1.00         +       0min       Number of operator cha       Parameters       1.00         +       0.00       Extended impedance       Parameters       1.00         +       0.00       Extended impedance       Parameters       1.00         DeltaT =       Time difference between desired and actual departure or arrival time       Consider connections with DeltaT > 0, if connection with DeltaT = 0 exists         Impedance =       ibeCoefficient       Attribute       BoxCox       Lambda         1.00       PJT [min]       1.00       1.00       1.00         +       1.00       DeltaT(early) [min]       1.00       1.00         +       1.00       DeltaT(late) [min]       1.00       1.00	+	1.00	Walk time					1.00		
+       1.00       Transfer wait time       Parameters       □       1.00         +       2min       Number of transfers       *       Formula       □       1.00         +       0min       Number of operator cha       Parameters       □       1.00         +       0.00       Extended impedance       Parameters       □       1.00         +       0.00       Extended impedance       Parameters       □       1.00         +       0.00       Extended impedance       Parameters       □       1.00         DeltaT =       Time difference between desired and actual departure or arrival time       □       Consider connections with DeltaT > 0, if connection with DeltaT = 0 exists         Impedance =       □       beCoefficient       Attribute       BoxCox       Lambda         1.00       PJT [min]       □       1.00       1.00       1.00       1.00         +       1.00       DeltaT(late) [min]       □       1.00       1.00       1.00         +       1.00       DeltaT(late) [min]       □       1.00       1.00       1.00	+	1.00	Origin wait time		Parameters			1.00		
+ 2min       Number of transfers       * Formula       □       1.00         + 0min       Number of operator cha       Parameters       □       1.00         + 0.00       Extended impedance       Parameters       □       1.00         + 0.00       Extended impedance       Parameters       □       1.00         DeltaT = Time difference between desired and actual departure or arrival time       ○       Consider connections with DeltaT > 0, if connection with DeltaT = 0 exists         Impedance =       •       •       •       •       1.00         •       0.00       Fare       1.00       •       1.00         + 0.00       Fare       1.00       •       1.00       •         + 1.00       DeltaT(early) [min]       □       1.00       •       1.00         + 1.00       DeltaT(late) [min]       □       1.00       •       •	+	1.00	Transfer wait time		Parameters			1.00		
+ 0min       Number of operator cha       Parameters       □       1.00         + 0.00       Extended impedance       Parameters       □       1.00         DeltaT = Time difference between desired and actual departure or arrival time       □       Consider connections with DeltaT > 0, if connection with DeltaT = 0 exists         Impedance =       beCoefficient       Attribute       BoxCox       Lambda         1.00       PJT [min]       □       1.00         + 0.00       Fare       □       1.00         + 1.00       DeltaT(early) [min]       □       1.00         + 1.00       DeltaT(late) [min]       □       1.00	+	2min	Number of transfers	*	Formula			1.00		
+       0.00       Extended impedance       Parameters       □       1.00         DeltaT = Time difference between desired and actual departure or arrival time       Occosider connections with DeltaT > 0, if connection with DeltaT = 0 exists         Impedance =         bbcCoefficient       Attribute       BoxCox       Lambda         1.00       PJT [min]       □       1.00         +       0.00       Fare       □       1.00         +       1.00       DeltaT(early) [min]       □       1.00         +       1.00       DeltaT(late) [min]       □       1.00	+	0min	Number of operator cha		Parameters			1.00		
DeltaT = Time difference between desired and actual departure or arrival time         Consider connections with DeltaT > 0, if connection with DeltaT = 0 exists         Impedance =         bbcCoefficient       Attribute         BoxCox       Lambda         1.00       PJT [min]         + 0.00       Fare         1.00       DeltaT(early) [min]         + 1.00       DeltaT(early) [min]         -       1.00	+	0.00	Extended impedance		Parameters			1.00		
Impedance =         be Coefficient       Attribute       BoxCox       Lambda         1.00       PJT [min]       1.00       1.00         +       0.00       Fare       1.00         +       1.00       DeltaT(early) [min]       1.00         +       1.00       DeltaT(late) [min]       1.00	Del	DeltaT = Time difference between desired and actual departure or arrival time Consider connections with DeltaT > 0, if connection with DeltaT = 0 exists								
beCoefficient         Attribute         BoxCox         Lambda           1.00         PJT [min]         1.00         1.00           +         0.00         Fare         1.00           +         1.00         DeltaT(early) [min]         1.00           +         1.00         DeltaT(late) [min]         1.00	Im	pedance =								
1.00       PJT [min]       1.00         +       0.00       Fare       1.00         +       1.00       DeltaT(early) [min]       1.00         +       1.00       DeltaT(late) [min]       1.00	nbe	Coefficient	Attribute				BoxCox	Lambda		
+       0.00       Fare       1.00         +       1.00       DeltaT(early) [min]       1.00         +       1.00       DeltaT(late) [min]       1.00		1.00	PJT [min]					1.00		
+ 1.00       DeltaT(early) [min]       1.00         + 1.00       DeltaT(late) [min]       1.00	+	0.00	Fare					1.00		
+ 1.00 DeltaT(late) [min]   1.00  0K Cancel	+	1.00	DeltaT(early) [min]					1.00		
OK Cancel	+	1.00	DeltaT(late) [min]					1.00		
OK Cancel										
OK Cancel										
								ОК	Cancel	

Task 10: List of rec	juired Matrices
----------------------	-----------------

Matrix fi	lter	М	atrix([NO] > 0)						
Count: 20	No	Code	Name	MatrixType	ObjectTypeRef	Sum	DSegCode	DataSourceType	Formula
1	1	IVT	In-vehicle time PT	Skim	Zone	2171186504783.352	PT	Data	
2	2	DIS	Trip distance C	Skim	Zone	209019199.482	С	Data	
3	3	TTC	tCur C	Skim	Zone	132143513.108	С	Data	
4	4	ACT	Access time PT	Skim	Zone	2170987743181.847	PT	Data	
5	5	NTR	Number of transfers PT	Skim	Zone	2170977502501.000	PT	Data	
6	6	JRD	Journey distance PT	Skim	Zone	2171110355434.503	PT	Data	
7	8	EGT	Egress time PT	Skim	Zone	2170987638257.237	PT	Data	
8	9	Distribution HA	Distribution HA	OD demand	Zone	17664.939		Data	
9	10	Distribution HO	Distribution HO	OD demand	Zone	856655.000		Data	
10	11	Distribution HW	Distribution HW	OD demand	Zone	503591.000		Data	
11	14		Moduswahl HA x C	OD demand	Zone	11063.559	С	Data	
12	15		Moduswahl HA x PuT	OD demand	Zone	6601.380	PT	Data	
13	16		Moduswahl HO x C	OD demand	Zone	786114.398	С	Data	
14	17		Moduswahl HO x PuT	OD demand	Zone	70540.602	PT	Data	
15	18		Moduswahl HW x C	OD demand	Zone	433406.393	С	Data	
16	19		Moduswahl HW x PuT	OD demand	Zone	70184.607	PT	Data	
17	20	TotalHomebased		OD demand	Zone	668369.538	С	Data	
18	21	TotalHomebased		OD demand	Zone	147326.588	PT	Data	
19	22	С	Car	OD demand	Zone	1347802.635	С	Data	
20	23	PT	PuT	OD demand	Zone	294653.177	PT	Data	



Task 11: Final network modeling

### Appendix C

Results of Mode Choice Analysis for Homebased-other trip







## Appendix D

Results of Mode Choice Analysis for Homebased-work trip







# Appendix E

Results of Public Transport Assignment in the scenario Erding Ring Closure with Scale Dimension of 10,000 PERS (Based on VISUM)

